UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Form 10-Q

[X] Quarterly report under Section 13 or 15(d) of the Securities Exchange Act of 1934

For The Quarterly Period Ended June 30, 2004

Commission File No. 0-9115

MATTHEWS INTERNATIONAL CORPORATION

(Exact Name of registrant as specified in its charter)

PENNSYLVANIA (State or other jurisdiction of incorporation or organization) 25-0644320 (I.R.S. Employer Identification No.)

TWO NORTHSHORE CENTER, PITTSBURGH, PA

15212-5851

(Address of principal executive offices)

(Zip Code)

Registrant's telephone number, including area code (412) 442-8200

NOT APPLICABLE

(Former name, former address and former fiscal year, if changed since last report)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15 (d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

Yes [X] No []

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act).

Yes [X] No []

As of July 31, 2004, shares of common stock outstanding were:

Class A Common Stock 32,267,330 shares

PART I - FINANCIAL INFORMATION MATTHEWS INTERNATIONAL CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEET (UNAUDITED)

<C>

<C>

(Dollar amounts in thousands, except per share data)

<TABLE>

<CAPTION>

June 30, 2004 September 30, 2003

<C>

<S>

ASSETS
Current assets:

Cash and cash equivalents Short-term investments Accounts receivable, net Inventories: Materials and finished goods Labor and overhead in process	2,859	4,588 62,883 \$25,576 1,489
Other current assets	30,743 3,751	27,065 4,564
Total current assets Investments Property, plant and equipment: Cost Less accumulated depreciation	226,750 7,650 147,967 (82,941)	166,054 4,561 140,487 (70,854)
Deferred income taxes and other assets Goodwill Other intangible assets, net	26, 156.847	69,633 061 32,182 154,690 13,062
Total assets	\$495,135	
LIABILITIES AND SHAREHOLDERS' ECCurrent liabilities: Long-term debt, current maturities Accounts payable Accrued compensation Accrued income taxes Customer prepayments Other current liabilities Total current liabilities	\$ 18,2 17,913 26,786 17,197	19,805 24,745
Long-term debt Estimated finishing costs Postretirement benefits Environmental reserve Deferred income taxes Other liabilities and deferred revenue	51,795 4,798 17,678 10,713 2,274 13,24	57,023 4,863 17,644 11,154 3,441 45 13,506
Shareholders' equity: Common stock Additional paid in capital Retained earnings Accumulated other comprehensive income Treasury stock, at cost	36,334 10,250 293,264 10,309 (57,053) 293,104	36,334 6,476 257,559 6,643 (50,784) 256,228
Total liabilities and shareholders' equity	\$495,	•

 | |

MATTHEWS INTERNATIONAL CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENT OF INCOME (UNAUDITED)

(Dollar amounts in thousands, except per share data)

<TABLE> <CAPTION>

	Three Mo June 30,	nths Ended	Nine Mo June 30,	nths Ended	
	2004	2003	2004 20	03	
<\$> Sales	<c> \$120,635</c>	<c> \$116,145</c>	-	C> \$340,799)
Cost of sales	(72,181	(71,696	(223,815) (214,757	7)
Gross profit	48,454	44,449	138,709	126,042	
Selling and administrative	expenses	(23,242)	(22,876)	(69,478)	(67,157)
Operating profit	25,21	 12 21,57	3 69,231	58,885	

Investment income Interest expense Other income (deductions), ne Minority interest	(613)	(578) (118)	4 1,09) (1,493 (84) 9) (3,98	3) (2, (203)	250) (129)
Income before income taxes	2	23,498	20,086	64,648	54,211
Income taxes		-	7) (25,0		1,037)
Net income			89 \$39,		\$33,174
==	===	\$.38	\$1.23 ===== \$1.21 ===== 75 \$.1	\$1.03 ===	 .0825

MATTHEWS INTERNATIONAL CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENT OF CASH FLOWS (UNAUDITED)

(Dollar amounts in thousands, except per share data)

<table></table>				
<caption></caption>				
	Ju	ne 30,	s Ended	
			2003	
<s></s>		>		
Cash flows from operating activities:				
Net income		39,564	\$3	3,174
Adjustments to reconcile net income				
provided by operating activities:				
Depreciation and amortization			11,138	11,044 339 (13,086) 2,555 (1,090) (4,395)
Change in deferred taxes		1	65	339
Changes in working capital items			7,526	(13,086)
Decrease in other assets		5,1	38	2,555
Decrease in estimated finishing cost	S		(65)	(1,090)
		(5	85)	(4,395)
Decrease in other liabilities Decrease in postretirement benefits Tax benefit of exercised stock option			(212)	(375)
Tax benefit of exercised stock optio	ns 2	2,805	5,10	00
Net gain on sales of assets		(1	33)	(336)
				,
Net cash provided by operating act				1 32,930
Cash flows from investing activities:				
Capital expenditures		(6,82)	26)	(6,761)
Proceeds from sales of assets			850	2,216
Purchases of investment securities			(15,193)	(145)
Capital expenditures Proceeds from sales of assets Purchases of investment securities Proceeds from disposition of investm	nent sec	urities		17 16
Net cash used in investing activitie			· (21 152)	(4,674)
Net easif used in investing activitie	,s 		21,132)	(4,074)
Cash flows from financing activities:				
Proceeds from long-term debt			52 066	_
Payments on long-term debt		G	45 907)	(31 495)
Proceeds from the sale of treasury st.	ock	(-	7 271	10.685
Proceeds from long-term debt Payments on long-term debt Proceeds from the sale of treasury st Purchases of treasury stock	OCK	(12	2 570)	(2 199)
Dividends	(3 859)	(2,6	(15)
Dividonds		رروی, 		.13)
Net cash used in financing activities	es		(2,999)	(25,624)
<i>g</i>				, , ,
Effect of exchange rate changes on ca	ash		1,33	7 4,205
			-	,

</TABLE>

MATTHEWS INTERNATIONAL CORPORATION AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS JUNE 30, 2004

(Dollar amounts in thousands, except per share data)

Note 1. Nature of Operations

Matthews International Corporation ("Matthews"), founded in 1850 and incorporated in Pennsylvania in 1902, is a designer, manufacturer and marketer principally of memorialization products, caskets and cremation equipment for the cemetery and funeral home industries; merchandising solutions; and custommade products which are used to identify people, places, products and events. The Company's products and operations are comprised of six business segments: Bronze, York Casket, Cremation, Graphics Imaging, Marking Products and, as of July 19,2004, Merchandising Solutions (see Note 13). The Bronze segment is a leading manufacturer of cast bronze memorials and other memorialization products, cast and etched architectural products and is a leading builder of mausoleums in the United States. The York Casket segment is a leading casket manufacturer in the United States and produces a wide variety of wood and metal caskets. The Cremation segment is a leading designer and manufacturer of cremation equipment and cremation caskets primarily in North America. The Graphics Imaging segment manufactures and provides printing plates, pre-press services and imaging services for the corrugated and flexible packaging industries. The Marking Products segment designs, manufactures and distributes a wide range of marking equipment and consumables for identifying various consumer and industrial products, components and packaging containers.

The Company has manufacturing and marketing facilities in the United States, Australia, Canada and Europe.

Note 2. Basis of Presentation

The accompanying consolidated financial statements have been prepared in accordance with generally accepted accounting principles for interim financial information for commercial and industrial companies and the instructions to Form 10-Q and Rule 10-01 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for fair presentation have been included. Operating results for the three months and nine months ended June 30, 2004 are not necessarily indicative of the results that may be expected for the fiscal year ending September 30, 2004. For further information, refer to the consolidated financial statements and footnotes thereto included in the Company's Annual Report on Form 10-K for the year ended September 30, 2003.

The consolidated financial statements include all majority-owned foreign and domestic subsidiaries. The consolidated financial statements also include the accounts of the Company's 50%-owned affiliate, S+T GmbH & Co. KG. All intercompany accounts and transactions have been eliminated.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS, continued (Dollar amounts in thousands, except per share data)

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Certain reclassifications have been made in the prior period financial statements to conform to the current period presentation.

Note 3. Stock-Based Compensation

The Company has accounted for its stock-based compensation plans in accordance with the intrinsic value provisions of Accounting Principles Board Opinion No. 25, "Accounting for Stock Issued to Employees." Accordingly, the Company did not record any compensation expense in the consolidated financial statements for its stock-based compensation plans. In accordance with Statement of Financial Accounting Standards ("SFAS") No. 148, "Accounting for Stock-Based Compensation-Transition and Disclosure", the following table illustrates the effect on net income and earnings per share had compensation expense been recognized consistent with the fair value provisions of SFAS No. 123, "Accounting for Stock-Based Compensation."

<TABLE>

<CAPTION>

	Three Mor June 3	nths Ended 0,	June	Nine Months E	nded
	2004	2003	2004	2003	
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	
Net income, as reported	9	\$14,380	\$12,289	\$39,564	\$33,174
Net income, pro forma		13,939	11,919	38,359	32,175
Basic earnings per share,	as reported	.45	.38	1.23	1.05
Diluted earnings per shar	e, as report	ed .44	.38	1.21	1.03
Basic earnings per share,	pro forma	.43	.37	1.19	1.02
Diluted earnings per shar	e, pro form	a .43	.37	1.18	1.01

</TABLE>

Note 4. Income Taxes

Income tax provisions for the Company's interim periods are based on the effective income tax rate expected to be applicable for the full year. The difference between the estimated effective tax rate for fiscal 2004 of 38.8% and the Federal statutory rate of 35% primarily reflects the impact of state and foreign income taxes.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS, continued (Dollar amounts in thousands, except per share data)

Note 5. Earnings Per Share

<TABLE>

<CAPTION>

CIN TIOT	Three M June	Ionths Ended	d I	Ended		
	2004	2003	2004	2003	-	
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>		
Net income	\$1	4,380 \$1	12,289	\$39,564	\$33,174 	

Weighted-average common

shares outstanding 32,252,258 31,845,791 32,172,931 31,523,848

Dilutive securities, primarily stock options	466,418	497,032	486,574	664,	760
Diluted weighted-average common shares outstanding	32,718,676	5 32,342,8 ======	323 32,6 ======	59,505	32,188,608
Basic earnings per share	\$.45	\$.38	\$1.23	\$1.05 =	
Diluted earnings per share ====	\$.44 ====	\$.38	\$1.21	\$1.03	

 | | | | |

Note 6. Segment Information

The Company's products and operations are comprised of six business segments: Bronze, York Casket, Cremation, Graphics Imaging, Marking Products and, as of July 19, 2004, Merchandising Solutions (see Note 13). The segments are described under Nature of Operations (Note 1). Management evaluates segment performance based on operating profit (before income taxes) and does not allocate non-operating items such as investment income, interest expense, other income (deductions), net and minority interest.

Information about the Company's segments follows:

<TABLE>

<CAPTION>

	Three Months Ended June 30,			Nine N ne 30,	s Ended	
	2004	2003	3 20	004	2003	3
<s></s>	<c></c>	<c></c>	<c></c>	> <	<c></c>	
Sales to external	customers:					
Bronze	\$51,6	566 \$	549,978	\$144,2	71	\$137,699
York Casket	27	,013	28,661	91,1	55	92,437
Cremation	5,1	63	4,374	16,552	2	15,052
Graphics Imagin	ng	27,495	24,718	82	2,420	71,705
Marking Produc	ets	9,298	8,414	28,	126	23,906
	\$120.635	\$11 <i>6</i>	 5,145	 \$362,524	 1	- 8340.799
	#120,033	•	J,17J =======	, .	r 4	

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS, continued (Dollar amounts in thousands, except per share data)

<CAPTION>

	Three Mo June 3			N June 3	Nine Mo 30,	nths En	ded
-	2004	200	03	2004	2	003	
<s></s>	<c></c>	<c></c>	·	<c></c>	<(">	
Operating profit:							
Bronze	\$14,9	48	\$14,278	5	\$35,338	\$3:	5,461
York Casket	3,	763	2,863		13,390	10,	013
Cremation	(1	28)	175		829	887	
Graphics Imaging	g	4,959	3,17	8	14,63	4	9,506
Marking Product	S	1,670	1,07	9	5,040) 3	3,018
	\$25,212	\$2	 1,573	\$69,	,231	\$58,88	35
=		=== :					===

</TABLE>

Note 7. Comprehensive Income

Comprehensive income consists of net income adjusted for changes, net of tax,

in cumulative foreign currency translation, unrealized gains and losses on investments, unrealized gains and losses on a derivative instrument and minimum pension liability. For the three months ended June 30, 2004 and 2003, comprehensive income was \$13,672 and \$16,878, respectively. For the nine months ended June 30, 2004 and 2003, comprehensive income was \$43,230 and \$43,628, respectively.

Note 8. Goodwill and Other Intangible Assets

Under Statement of Financial Accounting Standards ("SFAS") No. 142, "Goodwill and Other Intangible Assets," goodwill is no longer amortized but is subject to annual review for impairment. In general, when the carrying value of a reporting unit exceeds its implied fair value, an impairment loss must be recognized. For purposes of testing for impairment the Company uses a combination of valuation techniques, including discounted cash flows. Intangible assets are amortized over their estimated useful lives unless such lives are considered to be indefinite. The Company performs its annual impairment review in its second fiscal quarter.

Changes to goodwill, net of accumulated amortization, for the nine months ended June 30, 2004, are as follows.

<TABLE>

CAI HOW							
	York		Graphics	Markir	ng		
Bro	nze Ca	sket Cr	emation	Imaging	Products	s Consolio	lated
<s> <</s>	C> <	<c> <</c>	:C> <	:C> <	<c></c>	<c></c>	
Balance at September 30,	2003	\$ 72,122	\$ 40,700	6 \$ 6,530	6 \$ 35,16	51 \$ 165	\$154,690
Additions during period	-	-	-		-		
Translation and							
other adjustments	985	5 -	-	1,172	- 2	2,157	
Balance at June 30, 2004	\$ 7	3,107 \$	40,706	\$ 6,536	\$ 36,333	\$ 165	\$156,847

</TABLE>

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS, continued (Dollar amounts in thousands, except per share data)

Cremation Segment

The following tables summarize the carrying amounts and related accumulated amortization for intangible assets as of June 30, 2004 and September 30, 2003, respectively.

	, .			, .	Accumulated Amortization
June 30, 2004:					
Trade names Customer relati Copyrights/pate	onships ents/other	4,100	(623) (224)	300	-
September 30,					
Trade names Customer relati Copyrights/pate	\$ 8,00 onships	4,100	(442)	-	-
	\$13,400	\$(601)	\$ 3	300	8(37)

York Casket Segment

^{*} Not subject to amortization

Intangible assets established for customer relationships and copyrights, patents and other are amortized over their estimated useful lives of 17 years and 15 years, respectively. For the three-month period ended June 30, 2004, amortization expense on intangible assets was \$82 for the York Casket segment and \$5 for the Cremation segment. For the nine-month period ended June 30, 2004, amortization expense was \$246 for the York Casket segment and \$15 for the Cremation segment. Amortization expense on intangible assets is expected to approximate \$350 each year between 2004 and 2008.

Note 9. Long-term Debt

On December 3, 2001, the Company entered into a Revolving Credit Facility for \$125.0 million with a syndicate of financial institutions. The facility was scheduled to mature on November 30, 2004. On April 21, 2004, the Company signed an amendment to the facility which extended its maturity to April 30, 2009. Borrowings under the amended facility bear interest at LIBOR plus a factor ranging from .50% to 1.00% based on the Company's leverage ratio. The leverage ratio is defined as net indebtedness divided by EBITDA (earnings before interest, taxes, depreciation and amortization). The Company is required to pay an annual commitment fee ranging from .20% to .30% (based on the Company's leverage ratio) of the unused portion of the facility. The Revolving Credit Facility, as amended, requires the Company to maintain certain leverage and interest coverage ratios. A portion of the facility (not to exceed \$10.0 million) is available for the issuance of trade and standby letters of credit.

Effective April 30, 2004, the Company increased its outstanding borrowings under the facility to \$50.0 million and simultaneously entered into an interest rate swap that fixed the interest rate on such borrowings at 3.16% for a five-year period. The interest rate swap has been designated as a cash NOTES TO CONSOLIDATED FINANCIAL STATEMENTS, continued (Dollar amounts in thousands, except per share data)

flow hedge of the future variable interest payments under the Revolving Credit Facility which are considered probable of occurring. Based on the Company's assessment, all of the critical terms of the hedge match the underlying terms of the hedged debt and related forecasted interest payments and as such, these hedges are considered effective.

The fair value of the interest rate swap reflected an unrealized gain of \$934,000 (\$570,000 after tax) at June 30, 2004 that is included in equity as part of accumulated other comprehensive income. Assuming market rates remain constant with the rates at June 30, 2004, approximately \$115,000 of the \$570,000 gain included in accumulated other comprehensive income is expected to be recognized in earnings over the next 12 months.

At June 30, 2004 the outstanding balance on the Revolving Credit Facility was \$50,000 and the weighted-average interest rate on the outstanding borrowings under this facility was 3.16%. Equal quarterly payments of \$2,500 plus interest are due on the facility until its maturity in April 2009.

Caggiati S.p.A. has financed several acquisitions through loans with various Italian banks. Outstanding borrowings on these loans totaled U.S.\$14,800 at June 30, 2004. Caggiati also has four lines of credit totaling approximately U.S.\$12,000 with the same Italian banks. Outstanding borrowings on these lines approximated \$4,500 at June 30, 2004.

Note 10. Pension and Other Postretirement Benefit Plans

The Company provides defined benefit pension and other postretirement plans to certain employees. The following represents the net periodic pension and other postretirement benefit cost (income) for the plans in accordance with the revised version of SFAS No. 132, "Employer's Disclosures about Pensions and Other Postretirement Benefits," as described in Note 12:

<TABLE>

<CAPTION>

	Pension	Other Postre						
Three months ended June 30, 2004 2003 2004 20								
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>				
Service cost	\$906	\$759	\$99	\$72				
Interest cost	1,266	1,208	262	265				
Expected return	n on							

plan assets	(1,484)	(1,154)	-	-
Amortization:				
Prior service cost	28	28	(322)	(322)
Net actuarial loss	293	313	112	124
Net benefit cost	\$1,009	\$ 1,154	\$151	\$139

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS, continued (Dollar amounts in thousands, except per share data)

<ca< td=""><td>רת</td><td>ГΤ</td><td>\cap</td><td>Τ∕.</td></ca<>	רת	ГΤ	\cap	Τ∕.
\sim UA	r	יוו	יועו	·

	Pension		Other Posti		
Nine months ended	June 30,	2004	2003	2004	2003
<s></s>	<c> <</c>	:C>	<c></c>	<c></c>	
Service cost	\$ 2,719	\$2,277	7 \$2	96 \$2	216
Interest cost	3,797	3,624	786	795	
Expected return on					
plan assets	(4,452)	(3,462)	-	-	
Amortization:					
Prior service cost	84	84	(966)	(966)
Net actuarial loss	879	939	33:	5 372	2
Net benefit cost	\$ 3,027	\$3,46	52 \$	451 \$	3417

</TABLE>

Benefit payments under the Company's principal retirement plan are made from plan assets, while benefit payments under the supplemental retirement plan and postretirement benefit plan are made from the Company's operating funds. Due to the IRS full funding limitations, the Company is not required to make any contributions to its principal retirement plan in fiscal year 2004. As of June 30, 2004, contributions of \$258 and \$588 have been made under the supplemental retirement plan and postretirement plan, respectively. The Company currently anticipates contributing an additional \$96 and \$338 under the supplemental retirement plan and postretirement plan, respectively, for the remainder of fiscal 2004.

Note 11. Acquisitions

In August 2003, Matthews acquired Reproservice Eurodigital GmbH Munchen ("Reproservice Munich"), a German graphics and flexographic printing plate manufacturer in Munich, Germany. The transaction was structured as a stock purchase, at an acquisition price of 4.1 million Euros (U.S.\$4,800). Products and services of Reproservice Munich include pre-press packaging, digital and analog flexographic printing plates, design, art work, lithography and color separation. The combination of Matthews and Reproservice Munich is an important part of the Matthews strategy to increase its European presence in the graphics industry.

In May 1998, Matthews acquired a 50% interest in O.N.E. Color Communications ("O.N.E."), a digital graphics service company located in Oakland, California. The purchase price consisted of \$2,000 cash upon closing plus an additional \$2,750 in 2001, which was based upon the attainment of certain operating performance levels of O.N.E. The purchase agreement also required Matthews to acquire the remaining 50% interest no later than May 2004, with the purchase price contingent on the attainment of certain operating performance levels of O.N.E., but not less than \$4,500. The accounts of O.N.E. have been included in the consolidated financial statements of Matthews since May 1998 and a

liability was recorded for the future minimum payout. Effective July 31, 2003, Matthews completed the purchase of the remaining 50% interest in O.N.E. for \$5,700.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS, continued (Dollar amounts in thousands, except per share data)

Note 12. Accounting Pronouncements

In January 2003, the FASB issued Interpretation No. 46, "Consolidation of Variable Interest Entities" ("FIN 46"). This interpretation addresses consolidation by business enterprises of variable interest entities with certain characteristics. FIN 46 is effective immediately for all variable interest entities created after January 31, 2003. In October 2003, the FASB agreed to defer the effective date of FIN 46 for variable interest entities held by public companies that were acquired before February 1, 2003. The deferral will require that public companies adopt the provisions of FIN 46 for periods ending after December 15, 2003. The adoption of FIN 46 did not have a material impact on the Company's consolidated financial position and results of operations.

In July 2003, the EITF issued Issue No. 00-21 "Revenue Arrangements with Multiple Deliverables." Issue No. 00-21 addresses certain aspects of the accounting by a vendor for arrangements under which it will perform multiple revenue generating activities. The provisions of Issue No. 00-21 were effective July 1, 2003, and have been applied prospectively by the Company to the finishing and storage elements of its pre-need sales.

In December 2003, the FASB issued a revised version of SFAS No. 132 "Employer's Disclosures about Pensions and Other Postretirement Benefits." This statement requires additional disclosures about assets, obligations, cash flows and net periodic benefit costs of defined benefit plans and other defined benefit postretirement plans. The disclosure requirements are effective for annual financial statements with fiscal years ending after December 15, 2003 and for interim periods beginning after December 15, 2003. The disclosure requirements of the revised version of SFAS No. 132 have been adopted by the Company and are included in this Quarterly Report on Form 10-Q.

In May 2004, the FASB issued FASB Staff Position No. 106-2 ("FSP 106-2"), "Accounting and Disclosure Requirements Related to the Medicare Prescription Drug, Improvement and Modernization Act of 2003" (the "Act"). The Act introduces a prescription drug benefit under Medicare as well as a federal subsidy to sponsors of post-retirement health care benefit plans that provide a benefit that is at least actuarially equivalent to Medicare Part D. When adopted, FSP 106-2 will supersede FSP 106-1, "Accounting and Disclosure Requirements Related to the Medicare Prescription Drug, Improvement and Modernization Act of 2003," which was issued in January 2004 and permitted a sponsor of a post-retirement health care plan that provides a prescription drug benefit to make a one-time election to defer accounting for the effects of the Act until more authoritative guidance on the accounting for the federal subsidy was issued. The Company elected the one-time deferral allowed under FSP 106-1 and, as a result, any measures of the accumulated post-retirement benefit obligation or net periodic post-retirement benefit cost in the financial statements or accompanying notes do not reflect the effects of the Act on post-retirement health care benefit plans. FSP 106-2 provides authoritative guidance on the accounting for the federal subsidy and specifies the disclosure requirements for employers who have adopted FSP 106-2, including those who are unable to determine whether benefits provided under its plan are actuarially equivalent to Medicare Part D. FSP 106-2 is effective for the Company's fourth quarter of fiscal 2004, but it is not expected to have a material impact on the Company's consolidated financial position or results of operations.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS, continued (Dollar amounts in thousands, except per share data)

In July 2004, the Company acquired Cloverleaf Group, Inc. ("Cloverleaf"), a provider of merchandising solutions. Cloverleaf was formed by the recent merger of iDL, Inc., which is a merchandising solutions company headquartered near Pittsburgh, PA and Big Red Rooster, which is a marketing and design services organization located in Columbus, OH. The transaction was structured as an asset purchase, at a cost of approximately \$34,000. The transaction was also structured to include potential additional consideration during the next six years contingent on the future growth in value of the acquired operations. The acquisition is designed to expand the Company's products and services into the merchandising solutions market.

In July 2004, the Company acquired Holjeron Corporation, an industrial controls manufacturer located in Wilsonville, OR. The transaction was structured as a stock purchase, and is a part of Matthews' strategy to increase its presence in the marking products industry.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Cautionary Statement:

The following discussion should be read in conjunction with the consolidated financial statements of Matthews International Corporation and related notes thereto included in this Quarterly Report on Form 10-Q and the Company's Annual Report on Form 10-K for the year ended September 30, 2003. Any forward-looking statements contained herein are included pursuant to the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995. Such forward-looking statements involve known and unknown risks and uncertainties that may cause the Company's actual results in future periods to be materially different from management's expectations. Although the Company believes that the expectations reflected in such forward-looking statements are reasonable, no assurance can be given that such expectations will prove correct. Factors that could cause the Company's results to differ materially from the results discussed in such forward-looking statements principally include changes in domestic or international economic conditions, changes in raw material prices, changes in death rates, changes in foreign currency translation rates, changes in product demand or pricing as a result of consolidation in the industries in which the Company operates, changes in product demand or pricing as a result of domestic or foreign competitive pressures, unknown risks in connection with the Company's acquisitions, and technological factors beyond the Company's control.

Results Of Operations:

The following table sets forth certain income statement data of the Company expressed as a percentage of net sales for the periods indicated.

	Nine m	onths en	ided	Years ended		
	June 30,		September 30,			
	2004	2003	2003(1)	2002 2001	(2)	
Sales				0.0% 100.0%		
Gross profit Operating profit				1 37.5 42. 7.5 15.9 13		
Income before taxe Net income (3)		17.8	15.9	16.0 14.6 8 8.9 11.	18.2	
(-)					_	

- (1) The fourth quarter of fiscal 2003 included a net pre-tax charge of approximately \$1.0 million from special items which consisted of a pre-tax gain of \$2.6 million on the sale of a facility and a goodwill impairment charge of \$3.6 million.
- (2) Fiscal 2001 included pre-tax income of \$500,000 from special items, which consisted of a pre-tax gain of \$7.1 million on the sale of a subsidiary and asset impairments, restructuring costs and other special pre-tax charges totaling \$6.6 million.
- (3) Before cumulative effect of change in accounting. Fiscal 2002 reflected a pre-tax charge of \$5.3 million (\$.10 per share after tax) for transitional goodwill impairment.

Sales for the nine months ended June 30, 2004 were \$362.5 million and were \$21.7 million, or 6.4%, higher than sales of \$340.8 million for the nine months ended June 30, 2003. Bronze segment sales for the first nine months of fiscal 2004 were \$144.3 million compared to \$137.7 million for the first

Results Of Operations, continued:

nine months of fiscal 2003. The higher level of Bronze sales principally reflected the favorable impact of increases in the values of foreign currencies against the U.S. dollar and the effect of a temporary price surcharge instituted in April 2004 in response to increases in the cost of bronze ingot. These increases were offset partially by a decline in mausoleums sales. Sales for the York Casket segment were \$91.1 million for the first nine months of fiscal 2004 compared to \$92.4 million for the same period last year. The decrease primarily reflected the divestiture of a small manufacturing facility and several distribution operations in fiscal 2003. Sales for the Cremation segment were \$16.6 million for the first nine months of fiscal 2004 compared to \$15.1 million for the same period a year ago. The increase reflected higher sales volume of cremation equipment and cremation caskets compared to the same period a year ago. Sales for the Graphics Imaging segment in the first nine months of fiscal 2004 were \$82.4 million, compared to \$71.7 million for the same period a year ago. The increase reflected the acquisition of Reproservice Eurodigital GmbH Munchen ("Reproservice Munich") in August 2003, higher sales in the segment's other European operations and an increase in the value of the Euro against the U.S. dollar. These increases were partially offset by lower sales in the segment's domestic operations, which primarily related to the closure, in September 2003, of an unprofitable manufacturing business in North Carolina. Marking Products segment sales for the nine months ended June 30, 2004 were \$28.1 million, compared to \$23.9 million for the first nine months of fiscal 2003. The increase of \$4.2 million, or 17.7%, was principally due to higher volume, reflecting higher demand in North America and in Europe as a result of improving economic conditions, and the increase in value of the Swedish Krona against the U.S. dollar. Higher foreign currency values against the U.S. dollar had a favorable impact of approximately \$11.0 million on the Company's consolidated sales for the nine months ended June 30, 2004, compared to the same period a year ago.

Gross profit for the nine months ended June 30, 2004 was \$138.7 million, compared to \$126.0 million for the nine months ended June 30, 2003. Consolidated gross profit as of percent of sales increased from 37.0% for the first nine months of fiscal 2003 to 38.3% for the first nine months of fiscal 2004. The increase in consolidated gross profit and gross profit percentage primarily resulted from higher sales in the Marking Products and European Graphics Imaging businesses and manufacturing efficiency improvements in several of the Company's segments.

Selling and administrative expenses for the nine months ended June 30, 2004 were \$69.5 million, compared to \$67.2 million for the first nine months of fiscal 2003. The increase resulted from higher sales, the acquisition of Reproservice Munich in August 2003 and the increase in values of foreign currencies against the U.S. dollar. Consolidated selling and administrative expenses as a percent of sales were 19.2% for the nine months ended June 30, 2004 compared to 19.7% for the same period last year.

Operating profit for the nine months ended June 30, 2004 was \$69.2 million, representing an increase of \$10.3 million, or 17.6%, over operating profit of \$58.9 million for the nine months ended June 30, 2003. Bronze segment operating profit for the first nine months of fiscal 2004 was \$35.4 million, compared to \$35.5 million for the first nine months of fiscal 2003. The slight decrease reflected the absence of a one-time favorable adjustment recorded in fiscal 2003 related to the segment's pre-need memorial finishing cost liability, early retirement and severance costs incurred in March 2004, and increased bronze ingot costs. Bronze ingot costs at June 30, 2004 were approximately 60% higher than a year ago. These increases were partially offset by favorable impact in the third quarter of March 2004 personnel

Results Of Operations, continued:

reduction initiatives and the favorable impact of increases in the values of foreign currencies against the U.S. dollar. Operating profit for the York

Casket segment for the first nine months of fiscal 2004 was \$13.4 million, an increase of \$3.4 million, or 33.7%, over the same period a year ago. The increase reflected the favorable impact of the divestiture of unprofitable manufacturing and distribution operations during fiscal 2003, operating efficiencies realized in connection with productivity and quality initiatives at several of the segment's manufacturing facilities and a reduction in administrative expenses. Cremation segment operating profit was \$829,000 for the first nine months of fiscal 2004 compared to \$887,000 for the same period in fiscal 2003. The decrease primarily reflected the benefit of higher sales offset by changes in employee benefit costs in fiscal 2004 compared to fiscal 2003. Graphics Imaging operating profit for the nine months ended June 30, 2004 was \$14.6 million compared to \$9.5 million for the nine months ended June 30, 2003, an increase of 53.9%. The segment's operating profit was favorably impacted by the acquisition of Reproservice Munich, sales growth in the Company's other European operations, an increase in the value of the Euro against the U.S. dollar and the results of cost structure initiatives in the segment's domestic operations. Operating profit for the Marking Products segment for the first nine months of fiscal 2004 was \$5.0 million. representing an increase of \$2.0 million, or 67.0%, over the same period a year ago. The increase resulted from higher sales combined with an increase in the value of the Swedish Krona against the U.S. dollar. Higher foreign currency values against the U.S. dollar had a favorable impact of approximately \$2.5 million on the Company's consolidated operating profit for the nine months ended June 30, 2004 compared to the same period a year ago.

Investment income for the nine months ended June 30, 2004 was \$1.1 million, compared to \$985,000 for the nine months ended June 30, 2003. The increase over the prior period reflected higher levels of invested cash. Interest expense for the first nine months of fiscal 2004 was \$1.5 million, compared to \$2.3 million for the same period last year. The decline in interest expense reflected a lower average level of debt during the fiscal 2004 nine-month period combined with a reduction in the average borrowing rate.

Other deductions, net, for the nine months ended June 30, 2004 was \$203,000 compared to \$129,000 for same period last year. Minority interest deduction for the first nine months of fiscal 2004 was \$4.0 million, compared to \$3.3 million for the first nine months of fiscal 2003. The higher minority interest deduction for fiscal 2004 resulted from operating income growth in the Company's four European Graphics Imaging businesses that are not wholly owned.

The Company's effective tax rate for the nine months ended June 30, 2004 was 38.8%, which remained unchanged from the effective rate of 38.8% for the fiscal year ended September 30, 2003. The difference between the Company's effective tax rate and the Federal statutory rate of 35% primarily reflected the impact of state and foreign income taxes.

Goodwill:

Under Statement of Financial Standards ("SFAS") No. 142, "Goodwill and Other Intangible Assets", goodwill related to business combinations is no longer amortized, but is subject to periodic review for impairment. In general, when the carrying value of a reporting unit exceeds its implied fair value, an impairment loss must be recognized. For purposes of testing for impairment, the Company uses a combination of valuation techniques, including discounted cash flows. The Company performed its annual impairment review in the second

Results Of Operations, continued:

quarter of fiscal 2004 and determined that no adjustments to the carrying values of goodwill were necessary at that time.

Liquidity And Capital Resources:

Net cash provided by operating activities was \$65.3 million for the nine months ended June 30, 2004, compared to \$32.9 million for the first nine months of fiscal 2003. Operating cash flow for the first nine months of fiscal 2004 reflected net income adjusted for depreciation and amortization (non-cash charges), an increase in accrued income taxes and a tax benefit of \$2.8 million from exercised stock options. Accrued income taxes at June 30, 2004 were higher due to the current provision and overpayments of fiscal 2003

taxes that have been applied to future periods. For the nine months ended June 30, 2003, operating cash flow primarily reflected net income adjusted for depreciation, amortization, payments to customers under York Casket segment's rebate programs, a \$7.5 million contribution to the Company's pension plan, and a tax benefit of \$5.1 million from exercised stock options.

Cash used in investing activities was \$21.2 million for the nine months ended June 30, 2004, compared to \$4.7 million for the nine months ended June 30, 2003. Investing activities for the first nine months of fiscal 2004 included capital expenditures of \$6.8 million and net purchases of investments of \$15.2 million, partially offset by proceeds of \$850,000 from the sale of assets. Investing activities for the first nine months of fiscal 2003 primarily included capital expenditures of \$6.8 million, which was partially offset by proceeds of \$2.2 million from the sale of assets.

Capital expenditures reflected reinvestment in the Company's business segments and were made primarily for the purchase of new manufacturing machinery, equipment and facilities designed to improve product quality, increase manufacturing efficiency, lower production costs and meet regulatory requirements. Capital expenditures for the last three fiscal years were primarily financed through operating cash. Capital spending for property, plant and equipment has averaged \$8.9 million for the last three fiscal years. The capital budget for fiscal 2004 is \$13.9 million. The Company expects to generate sufficient cash from operations to fund all anticipated capital spending projects.

Cash used in financing activities for the nine months ended June 30, 2004 was \$3.0 million, reflecting repayment of long-term debt of \$45.9 million, stock repurchases of \$12.6 million and dividends of \$3.9 million to the Company's shareholders, partially offset by proceeds from long-term debt of \$52.1 million and net proceeds from the sale of treasury stock of \$7.3 million (stock option exercises). Cash used in financing activities for the nine months ended June 30, 2003 was \$25.6 million, reflecting payments on long-term debt of \$31.5 million, treasury stock purchases of \$2.2 million and dividends of \$2.6 million to the Company's shareholders. These payments were partially offset by proceeds of \$10.7 million from the sale of treasury stock (stock option exercises).

On December 3, 2001, the Company entered into a Revolving Credit Facility for \$125.0 million with a syndicate of financial institutions. The facility was scheduled to mature on November 30, 2004. On April 21, 2004, the Company signed an amendment to the facility which extended its maturity to April 30, 2009. Borrowings under the amended facility bear interest at LIBOR plus a factor ranging from .50% to 1.00% based on the Company's leverage ratio. The leverage ratio is defined as net indebtedness divided by EBITDA (earnings before interest, taxes, depreciation and amortization). The Company is

Liquidity and Capital Resources, continued:

required to pay an annual commitment fee ranging from .20% to .30% (based on the Company's leverage ratio) of the unused portion of the facility. The Revolving Credit Facility, as amended, requires the Company to maintain certain leverage and interest coverage ratios. A portion of the facility (not to exceed \$10.0 million) is available for the issuance of trade and standby letters of credit.

Effective April 30, 2004, the Company increased its outstanding borrowings under the facility to \$50.0 million and simultaneously entered into an interest rate swap that fixed the interest rate on such borrowings at 3.16% for a five-year period. The interest rate swap has been designated as a cash flow hedge of the future variable interest payments under the revolving credit facility.

At June 30, 2004 the outstanding balance on the Revolving Credit Facility was \$50.0 million and the weighted-average interest rate on the outstanding borrowings under this facility was 3.16%. Equal quarterly payments of \$2.5 million plus interest are due on the facility until its final maturity in April 2009.

Caggiati S.p.A. has four lines of credit totaling approximately U.S.\$12.0 million with various Italian banks. Outstanding borrowings on these lines approximated \$4.5 million at June 30, 2004.

The Company has a stock repurchase program, which was initiated in 1996. Under the program, the Company's Board of Directors has authorized the repurchase of a total of 10,000,000 shares (adjusted for stock splits) of Matthews common stock, of which 7,759,895 shares have been repurchased as of June 30, 2004. The buy-back program is designed to increase shareholder value, enlarge the Company's holdings of its common stock, and add to earnings per share. Repurchased shares may be retained in treasury, utilized for acquisitions, or reissued to employees or other purchasers, subject to the restrictions of the Company's Restated Articles of Incorporation.

Consolidated working capital of the Company was \$125.2 million at June 30, 2004, compared to \$89.7 million at September 30, 2003. Cash and cash equivalents were \$109.5 million at June 30, 2004, compared to \$67.0 million at September 30, 2003. The increase in working capital reflected the extension of the maturity on the Revolving Credit Facility. The increase in cash resulted from the Company's earnings during the current period and additional borrowings under its credit facility. The Company's current ratio was 2.2 at June 30, 2004 and September 30, 2003.

Environmental Matters:

The Company's operations are subject to various federal, state and local laws and regulations relating to the protection of the environment. These laws and regulations impose limitations on the discharge of materials into the environment and require the Company to obtain and operate in compliance with conditions of permits and other government authorizations. As such, the Company has developed policies and procedures with respect to environmental, safety and health, including the proper handling, storage and disposal of hazardous materials.

The Company is party to various environmental matters. These include obligations to investigate and mitigate the effects on the environment of the disposal of certain materials at various operating and non-operating sites. The Company is currently performing environmental assessments and remediation

Environmental Matters, continued:

at these sites, as appropriate. In addition, prior to its acquisition, York Casket was identified, along with others, by the Environmental Protection Agency as a potentially responsible party for remediation of a landfill site in York, PA. At this time, the Company has not been joined in any lawsuit or administrative order related to the site or its clean-up.

At June 30, 2004, an accrual of \$11.5 million was recorded for environmental remediation (of which \$786,000 has been classified in other current liabilities), representing management's best estimate of the probable and reasonably estimable costs of the Company's known remediation obligations. The accrual, which reflects previously established reserves assumed with the acquisition of York Casket and additional reserves recorded as a purchase accounting adjustment, does not consider the effects of inflation and anticipated expenditures are not discounted to their present value. While final resolution of these contingencies could result in costs different than current accruals, management believes the ultimate outcome will not have a significant effect on the Company's consolidated results of operations or financial position.

Acquisitions:

In July 2004, the Company acquired Cloverleaf Group, Inc. ("Cloverleaf"), a provider of merchandising solutions. Cloverleaf was formed by the recent merger of iDL, Inc., which is a merchandising solutions company headquartered near Pittsburgh, PA and Big Red Rooster, which is a marketing and design services organization located in Columbus, OH. The transaction was structured as an asset purchase, at a cost of approximately \$34.0 million. The transaction was also structured to include potential additional consideration during the next six years contingent on the future growth in value of the acquired operations. The acquisition is designed to expand the Company's products and services into the merchandising solutions market.

In July 2004, the Company acquired Holjeron Corporation, an industrial controls manufacturer located in Wilsonville, OR. The transaction was

structured as a stock purchase, and is a part of Matthews' strategy to increase its presence in the marking products industry.

In August 2003, Matthews acquired Reproservice Eurodigital GmbH Munchen ("Reproservice Munich"), a German graphics and flexographic printing plate manufacturer in Munich, Germany. The transaction was structured as a stock purchase, at an acquisition price of 4.1 million Euros (U.S.\$4.8 million). The combination of Matthews and Reproservice Munich is an important part of the Matthews strategy to increase its European presence in the graphics industry. Reproservice Munich, a family-owned business with annual sales of approximately U.S.\$6.0 million, was established in 1983. Products and services of Reproservice Munich include pre-press packaging, digital and analog flexographic printing plates, design, art work, lithography and color separation.

In May 1998, Matthews acquired a 50% interest in O.N.E. Color Communications ("O.N.E."), a digital graphics service company located in Oakland, California. The purchase price consisted of \$2.0 million cash upon closing plus an additional \$2.75 million in 2001, which was based upon the attainment of certain operating performance levels of O.N.E. The purchase agreement also required Matthews to acquire the remaining 50% interest no later than May 2004, with the purchase price contingent on the attainment of certain operating performance levels of O.N.E., but not less than \$4.5 million. The accounts of O.N.E. have been included in the consolidated financial statements

Acquisitions, continued:

of Matthews since May 1998 and a liability was recorded for the future minimum payout. Effective July 31, 2003, Matthews completed the purchase of the remaining 50% interest in O.N.E. for \$5.7 million.

Forward-Looking Information:

The Company's objective with respect to operating performance is to increase annual earnings per share in the range of 12% to 15% annually. For the past nine fiscal years, the Company has achieved an average annual increase in earnings per share of 15.2%. Matthews has a three-pronged strategy to attain the annual growth rate objective, which has remained unchanged from the prior year. This strategy consists of the following: internal growth (which includes productivity improvements, new product development and the expansion into new markets with existing products), acquisitions and share repurchases under the Company's stock repurchase program.

For the first nine months of fiscal 2004, the Company's earnings of \$1.21 per share were in line with management's original expectations for the year and represented an increase of 17.5% over earnings per share of \$1.03 for the same period last year. Based on the expected impact of the Company's recent acquisitions, anticipated internal growth and also considering recent significant increases in bronze ingot and steel costs, the Company expects to achieve diluted earnings per share of \$1.58 for the fiscal year ending September 30, 2004.

Critical Accounting Policies:

The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Therefore, the determination of estimates requires the exercise of judgment based on various assumptions and other factors such as historical experience, economic conditions, and in some cases, actuarial techniques. Actual results may differ from those estimates. A discussion of market risks affecting the Company can be found in "Quantitative and Qualitative Disclosures about Market Risk" in this Quarterly Report on Form 10-Q.

A summary of the Company's significant accounting policies are included in the Notes to Consolidated Financial Statements included in the Company's Annual Report on Form 10-K for the year ended September 30, 2003. Management believes that the application of these policies on a consistent basis enables the Company to provide useful and reliable financial information about the Company's operating results and financial condition. The following accounting policies involve significant estimates, which are considered critical to the

preparation of the Company's consolidated financial statements.

Allowance for Doubtful Accounts

The allowance for doubtful accounts is based on an evaluation of specific customer accounts in which available facts and circumstances indicate collectibility may be a problem. In addition, the allowance includes a general reserve for all customers based on historical collection experience.

Critical Accounting Policies, continued:

Long-Lived Assets

Property, plant and equipment, goodwill and other intangible assets are carried at cost. Depreciation on property, plant and equipment is computed primarily on the straight-line method over the estimated useful lives of the assets. Goodwill is no longer amortized, but is subject to periodic review for impairment. Intangible assets are amortized over their estimated useful lives, unless such lives are considered to be indefinite. A significant decline in cash flows generated from these assets may result in a write-down of the carrying values of the related assets.

Pension Costs

Pension assets and liabilities are determined on an actuarial basis and are affected by the market value of plan assets, estimates of the expected return on plan assets and the discount rate used to determine the present value of benefit obligations. Actual changes in the fair market value of plan assets and differences between the actual return on plan assets, the expected return on plan assets and changes in the selected discount rate will affect the amount of pension cost.

Environmental Reserve

Environmental liabilities are recorded when the Company's obligation is probable and reasonably estimable. Accruals for losses from environmental remediation obligations do not consider the effects of inflation, and anticipated expenditures are not discounted to their present value.

Revenue Recognition

Revenues are generally recognized when title and risk of loss pass to the customer, which is typically at the time of product shipment. For pre-need sales of memorials and vases, revenue is recognized when the memorial has been manufactured to the customer's specifications (e.g., name and birth date), title has been transferred to the customer and the memorial and vase are placed in storage for future delivery. A liability has been recorded in Estimated Finishing Costs for the estimated costs of finishing pre-need bronze memorials and vases that have been manufactured and placed in storage prior to July 1, 2003 for future delivery.

In July 2003, the Emerging Issues Task Force ("EITF") issued Issue No. 00-21 "Revenue Arrangements with Multiple Deliverables." Issue No. 00-21 addresses certain aspects of the accounting by a vendor for arrangements under which it will perform multiple revenue generating activities. The provisions of Issue No. 00-21 were effective July 1, 2003 and have been applied prospectively by the Company to the finishing and storage elements of its pre-need sales. Beginning July 1, 2003, revenue is deferred by the Company on the portion of pre-need sales attributable to the final finishing and storage of the pre-need merchandise. Deferred revenue for final finishing is recognized at the time the pre-need merchandise is finished and shipped to the customer. Deferred revenue related to storage is recognized on a straight-line basis over the estimated average time that pre-need merchandise is held in storage.

At June 30, 2004, the Company held 359,181 memorials and 236,844 vases in its storage facilities under the pre-need sales program.

Construction revenues are recognized under the percentage-of-completion method

of accounting using the cost-to-cost method. The Company offers rebates to certain customers participating in volume purchase programs. Rebates are

Critical Accounting Policies, continued:

estimated and recorded as a reduction in sales at the time the Company's products are sold.

LONG-TERM CONTRACTUAL OBLIGATIONS AND COMMITMENTS:

The following table summarizes the Company's contractual obligations at June 30, 2004, and the effect such obligations are expected to have on its liquidity and cash flows in future periods.

Long Term Contractual Obligations, continued:

<TABLE> <CAPTION>

Payments due in fiscal year: 2004 After Total (remainder) 2005 to 2006 2007 to 2008 2008 <S><C> <C> <C> <C> Contractual Cash Obligations: (Dollar amounts in thousands) \$50,000 \$2,500 Revolving credit facility \$20,000 \$20,000 \$7,500 Notes payable to banks 14,866 537 2,609 7,070 4,650 Short-term borrowings 4,536 4,536 Capital lease obligations 602 443 147 Non-cancelable operating leases 13,223 832 5,251 4,066 3,074 \$26,687 \$17,644 \$83,227 \$8,552 \$30,344 Total contractual cash obligations

</TABLE>

Benefit payments under the Company's principal retirement plan are made from plan assets, while benefit payments under the supplemental retirement plan and postretirement benefit plan are made from the Company's operating funds. Due to IRS full funding limitations, the Company is not required to make any contributions to its principal retirement plan in fiscal year 2004. As of June 30, 2004, contributions of \$258,000 and \$588,000 have been made under the supplemental retirement plan and postretirement plan, respectively. The Company currently anticipates contributing an additional \$96,000 and \$338,000 under the supplemental retirement plan and postretirement plan, respectively, for the remainder of fiscal 2004.

The Company believes that its current liquidity sources, combined with its operating cash flow and borrowing capacity, will be sufficient to meet its capital needs for the foreseeable future.

Accounting Pronouncements:

In January 2003, the FASB issued Interpretation No. 46, "Consolidation of Variable Interest Entities" ("FIN 46"). This interpretation addresses consolidation by business enterprises of variable interest entities with certain characteristics. FIN 46 is effective immediately for all variable interest entities created after January 31, 2003. In October 2003, the FASB agreed to defer the effective date of FIN 46 for variable interest entities held by public companies that were acquired before February 1, 2003. The deferral will require that public companies adopt the provisions of FIN 46 for periods ending after December 15, 2003. The adoption of FIN 46 did not have a material impact on the Company's consolidated financial position and results of operations.

Accounting Pronouncements, continued:

In December 2003, the FASB issued a revised version of SFAS No. 132 "Employer's Disclosures about Pensions and Other Postretirement Benefits."

This statement requires additional disclosures about assets, obligations, cash flows and net periodic benefit costs of defined benefit plans and other defined benefit postretirement plans. The disclosure requirements are effective for annual financial statements with fiscal years ending after December 15, 2003 and for interim periods beginning after December 15, 2003. The disclosure requirements of the revised version of SFAS No. 132 have been adopted by the Company and are included in this Quarterly Report on Form 10-Q.

In May 2004, the FASB issued FASB Staff Position No. 106-2 ("FSP 106-2"), "Accounting and Disclosure Requirements Related to the Medicare Prescription Drug, Improvement and Modernization Act of 2003" (the "Act"). The Act introduces a prescription drug benefit under Medicare as well as a federal subsidy to sponsors of post-retirement health care benefit plans that provide a benefit that is at least actuarially equivalent to Medicare Part D. When adopted, FSP 106-2 will supersede FSP 106-1, "Accounting and Disclosure Requirements Related to the Medicare Prescription Drug, Improvement and Modernization Act of 2003," which was issued in January 2004 and permitted a sponsor of a post-retirement health care plan that provides a prescription drug benefit to make a one-time election to defer accounting for the effects of the Act until more authoritative guidance on the accounting for the federal subsidy was issued. The Company elected the one-time deferral allowed under FSP 106-1 and, as a result, any measures of the accumulated post-retirement benefit obligation or net periodic post-retirement benefit cost in the financial statements or accompanying notes do not reflect the effects of the Act on post-retirement health care benefit plans. FSP 106-2 provides authoritative guidance on the accounting for the federal subsidy and specifies the disclosure requirements for employers who have adopted FSP 106-2, including those who are unable to determine whether benefits provided under its plan are actuarially equivalent to Medicare Part D. FSP 106-2 is effective for the Company's fourth quarter of fiscal 2004, but it is not expected to have a material impact on the Company's consolidated financial position or results of operations.

Item 3. Quantitative And Qualitative Disclosures About Market Risk

The following discussion about the Company's market risk involves forward-looking statements. Actual results could differ materially from those projected in the forward-looking statements. The Company has market risk related to changes in interest rates, commodity prices and foreign currency exchange rates. The Company does not generally use derivative financial instruments in connection with these market risks, except as noted below.

Interest Rates - The Company's most significant long-term debt instrument is the Revolving Credit Facility, as amended, which bears interest at variable rates based on LIBOR. Effective April 30, 2004, the Company increased its outstanding borrowings under the facility to \$50.0 million and simultaneously entered into an interest rate swap that fixed the interest rate on such borrowings at 3.16% for a five-year period. The interest rate swap has been designated as a cash flow hedge of the future variable interest payments under the Revolving Credit Facility. The fair value of the interest rate swap reflected an unrealized gain of \$934,000 (\$570,000 after tax) at June 30, 2004 that is included in equity as part of accumulated other comprehensive income. A decrease of 10% in market interest rates (i.e. a decrease from 3.5% to 3.15%) would result in a decrease of approximately \$360,000 in the fair value of the interest rate swap.

Quantitative And Qualitative Disclosures About Market Risk, continued:

Commodity Price Risks - In the normal course of business, the Company is exposed to commodity price fluctuations related to the purchases of certain materials and supplies (such as bronze ingot, steel and wood) used in its manufacturing operations. The Company obtains competitive prices for materials and supplies when available.

Foreign Currency Exchange Rates - The Company is subject to changes in various foreign currency exchange rates, including the Euro, the Canadian dollar, the Australian dollar and the Swedish Krona, in the conversion from local currencies to the U.S. dollar of the reported financial position and operating results of its non-U.S. based subsidiaries. An adverse change of 10% in exchange rates would have resulted in a decrease in sales of \$9.1 million and

a decrease in operating income of \$3.4 million for the nine months ended June 30, 2004.

Item 4. Controls and Procedures

Based on their evaluation at the end of the period covered by this Quarterly Report on Form 10-Q, the Company's chief executive officer and chief financial officer have concluded that the Company's disclosure controls and procedures (as defined in Rule 13a-15(e) under the Securities Exchange Act of 1934 (the "Exchange Act")) provide reasonable assurance that information required to be disclosed by the Company in reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in Securities and Exchange Commission rules and forms.

There have been no changes in the Company's internal controls over financial reporting that occurred during the fiscal quarter ended June 30, 2004 that have materially affected, or are reasonably likely to materially affect, the Company's internal controls over financial reporting.

PART II - OTHER INFORMATION

Item 2. Changes in Securities, Use of Proceeds, and Issuer Purchases of Equity Securities

Stock Repurchase Plan

The Company has a stock repurchase program, which was initiated in 1996. Under the program, the Company's Board of Directors has authorized the repurchase of a total of 10,000,000 shares (adjusted for stock splits) of Matthews common stock, of which 7,759,895 shares have been repurchased as of June 30, 2004. All purchases of the Company's common stock during the first nine months of fiscal 2004 were part of this repurchase program.

The following table shows the monthly fiscal 2004 stock repurchase activity:

<TABLE> <CAPTION>

	Total number of				
	Average shares purchased Maximum number				
	Total number price as part of a of shares that yet				
	of shares pa			ed may be purchased	
Period	purchased	share	plan	under the plan (1)	
<s></s>	<c> ·</c>	<c></c>	<c></c>	<c></c>	
October 2003	158,300	\$27.6	3 158,300	509,768	
November 2003	60,70	28.6	60,700	449,068	
December 2003	79	28.70	79	448,989	
January 2004	-	-	- 4	48,989	
February 2004	-	-	- 4	148,989	
March 2004	12,300	32.03	12,300	436,689	
April 2004	196,584	31.04	196,584	2,240,105	
May 2004	-	-	- 2,2	40,105	
June 2004	_	-	- 2,24	0,105	
			====		
Total	427,963	29.46	427,963		

<FN>

(1) In April 2004 the Company's Board of Directors authorized the purchase of an additional 2,000,000 shares of Matthews common stock, bringing the total authorization for stock repurchases to 10,000,000 shares.

</TABLE>

Item 4. Submission of Matters to a Vote of Security Holders

Item 6. Exhibits and Reports on Form 8-K

(a) Exhibits

Exhibit

No. Description

10.1 Asset Purchase Agreement Between I.D.L. Incorporated and Hugh Andrew, L.P. and Big Red Rooster, Inc. and Cloverleaf Group, L.P. and the iDL Shareholders and the BRR Shareholders and Cloverleaf Group, Inc. and Matthews International Corporation dated as of July 19, 2004.

31.1 Certification of Principal Executive Officer for David M.

- 31.2 Certification of Principal Financial Officer for Steven F. Nicola.
 - 32.1 Certification Pursuant to 18 U.S.C. Section 1350, as adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 for David M. Kelly.
 - 32.2 Certification Pursuant to 18 U.S.C. Section 1350, as adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 for Steven F. Nicola.

(b) Reports on Form 8-K

On April 26, 2004, Matthews filed a Current Report on Form 8-K under Item 9 in connection with a press release announcing the continuation of the Company's stock repurchase program and an increase in the authorization for stock repurchases by 2 million shares.

On April 21, 2004, Matthews filed a Current Report on Form 8-K under Item 12 in connection with a press release announcing its earnings for the quarter ended March 31, 2004.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date 8/10/04 David M. Kelly

David M. Kelly, Chairman of the Board,
President and Chief Executive Officer

Date 8/10/04 Steven F. Nicola

Steven F. Nicola, Chief Financial Officer,
Secretary and Treasurer

ASSET PURCHASE AGREEMENT

Between

I.D.L. Incorporated

and

Hugh Andrew, L.P.

and

Big Red Rooster, Inc.

and

Cloverleaf Group, L.P.

and

the iDL Shareholders

and

the BRR Shareholders

and

Cloverleaf Group, Inc.

and

Matthews International Corporation

rattie ws international Co

Dated as of July 19, 2004

TABLE OF CONTENTS

Page

ARTICLE I SALE AND PURCHASE OF THE ASSETS 1

- 1.1. Assets 1
- 1.2. Excluded Assets 3

ARTICLE II THE CLOSING 4

- 2.1. Place and Date 4
- 2.2. Purchase Price 4
- 2.3. Contingent Consideration 4
- 2.4. Closing Date Net Asset Statement 8
- 2.5. Assumption of Liabilities 8
- 2.6. Excluded Liabilities 9
- 2.7. Allocation of Purchase Price. 10

ARTICLE III REPRESENTATIONS AND WARRANTIES 10

- 3.1. Representations and Warranties of Seller 10
- 3.1.1. Authorization, etc. 10
- 3.1.2. Corporate Status 10
- 3.1.3. No Conflicts, etc. 11
- 3.1.4. Financial Statements 11
- 3.1.5. Absence of Undisclosed Liabilities 12
- 3.1.6. Taxes 12
- 3.1.7. Absence of Changes 13
- 3.1.8. Litigation 14
- 3.1.9. Compliance with Laws; Governmental Approvals and

Consents; Governmental Contracts 14

- 3.1.10. Operation of the Business 15
- 3.1.11. Assets 15
- 3.1.12. Contracts 15
- 3.1.13. Territorial Restrictions 16
- 3.1.14 Inventories 16
- 3.1.15. Customers 17
- 3.1.16. Suppliers; Raw Materials 17
- 3.1.17. Product Warranties 17
- 3.1.18. Absence of Certain Business Practices 17
- 3.1.19. Intellectual Property 17
- 3.1.20. Insurance 18
- 3.1.21. Real Property 18
- 3.1.22. Environmental Matters 19
- 3.1.23. Employees, Labor Matters, etc. 21
- 3.1.24. Employee Benefit Plans and Related Matters 21
- 3.1.25. Confidentiality 22
- 3.1.26. No Guarantees 23
- 3.1.27. Records 23
- 3.1.28. Brokers, Finders, etc. 23
- 3.1.29. Disclosure 23
- 3.2. Representations and Warranties of Buyer and Matthews 23
- 3.2.1. Corporate Status; Authorization, etc. 23

- 3.2.2. No Conflicts, etc. 24
- 3.2.3. Litigation 24
- 3.2.4. Brokers, Finders, etc. 24

ARTICLE IV COVENANTS 24

- 4.1. Covenants of Seller, iDL and BRR 24
- 4.1.1. Further Assurances 24
- 4.1.2. Liability for Transfer Taxes 25
- 4.1.3. Use of Business Name 25
- 4.14. Access to Records. 25
- 4.2. Covenants of Buyer. 25
- 4.2.1. Further Assurances 25
- 4.3. Operation of Business. 25
- 4.3.1. Operation of Business 25
- 4.3.2. Board of Directors 25
- 4.3.3. Board of Advisors 26
- 4.3.4. Newco Management Team 26
- 4.3.5. Approval Process 26
- 4.3.6. Financing and Investment Arrangements 27
- 4.3.7. BP Inventory 27
- 4.3.8. Acquisitions 27
- 4.4. Shareholder Net Worth. 27

ARTICLE V CLOSING DELIVERIES 27

- 5.1. Deliveries of Seller 27
- 5.1.1. Collateral Agreements. 27
- 5.1.2. Opinion of Counsel 28
- 5.1.3. Authorization 28
- 5.1.4. Transfer Documents 28
- 5.1.5. Consents and Estoppels 28
- 5.2. Deliveries of Buyer 28
- 5.2.1. Opinion of Counsel 28
- 5.2.2. Certified Resolutions 28
- 5.2.3. Collateral Agreements 28

ARTICLE VI EMPLOYEES AND EMPLOYEE BENEFIT PLANS 28

- 6.1. Employment of Seller's Employees 29
- 6.2. Employee Benefit Plans, Welfare and Fringe Benefit Plans 29
- 6.3. Bonus Arrangements. 30
- 6.4. Workers Compensation 30
- 6.5. Employment Taxes 30

ARTICLE VII DEFINITIONS, MISCELLANEOUS 31

- 7.1. Definition of Certain Terms 31
- 7.2. Indemnification 36
- 7.3. Survival of Representations and Warranties, etc. 40
- 7.4. Expenses 40
- 7.5. Severability 40
- 7.6. Notices 41
- 7.7. Miscellaneous 42
- 7.7.1. Headings 42
- 7.7.2. Entire Agreement 42
- 7.7.3. Counterparts 42
- 7.7.4. Governing Law, etc. 42
- 7.7.5. Binding Effect 42
- 7.7.6. Assignment 42
- 7.7.7. No Third Party Beneficiaries 42
- 7.7.8. Amendment; Waivers, etc. 43
- 7.7.9. Matters of Construction, Interpretation and the

Like. 43

SCHEDULES

SCHEDULE 1.2(a)(i) Excluded Cash

SCHEDULE 2.3(a)(iv) Free Cash Flow Example

SCHEDULE 2.3(c) Method of Distributing Contingent Consideration

SCHEDULE 2.5(j) Assumed Loans

SCHEDULE 4.3.5A Approval Process

SCHEDULE 4.3.5B Matthews Acquisition Policies

SCHEDULE 6.2 Assumed Employee Benefit Plans

Representations and Warranties

- -----

SCHEDULE 3.1.2(b) Qualifications to do Business

SCHEDULE 3.1.3 Conflicts

SCHEDULE 3.1.5 Undisclosed Liabilities

SCHEDULE 3.1.6(a) Contested Covered Taxes

SCHEDULE 3.1.6(b) Tax Extensions

SCHEDULE 3.1.6(c) Current Audits and Deficiencies

SCHEDULE 3.1.6(e) Tax Litigation

SCHEDULE 3.1.7 Absence of Changes

SCHEDULE 3.1.8 Litigation

SCHEDULE 3.1.9(a) Compliance with Laws

SCHEDULE 3.1.9(b) Governmental Approvals and other Consents

SCHEDULE 3.1.9(c) Governmental Contracts

SCHEDULE 3.1.10 Operation of the Business

SCHEDULE 3.1.11 Asset Exceptions

SCHEDULE 3.1.12(a) Contracts

SCHEDULE 3.1.12(c) Defaults and Consents under Contracts

SCHEDULE 3.1.14 Inventory Exceptions

SCHEDULE 3.1.15 Customers

SCHEDULE 3.1.16 Suppliers

SCHEDULE 3.1.17 Product Warranties

SCHEDULE 3.1.19(a) Owned Intellectual Property

SCHEDULE 3.1.19(d) Intellectual Property Licensing Arrangements

SCHEDULE 3.1.20 Insurance

SCHEDULE 3.1.21(a) Leases and Other Leases

SCHEDULE 3.1.22(a) Environmental Permits

SCHEDULE 3.1.22(d) Environmental Disclosure

SCHEDULE 3.1.23 Labor Matters

SCHEDULE 3.1.24(a) Employee Benefit Plans and Related Matters

SCHEDULE 3.1.25 Confidentiality

SCHEDULE 3.1.26 Guarantees

SCHEDULE 3.1.27 Records

SCHEDULE 3.1.30 Receivables

SCHEDULE 3.2.2 Governmental Approvals and other Consents

OTHER KEY CLOSING DOCUMENTS

Escrow Agreement

Employment Agreements - Murray and Beck

Employment Agreements - Key Employees

Transitional Lease Agreements

ASSET PURCHASE AGREEMENT

ASSET PURCHASE AGREEMENT, dated as of July 19, 2004 between Cloverleaf Group,

Inc., a Pennsylvania corporation (the "Buyer") and a wholly-owned subsidiary

of Matthews International Corporation, a Pennsylvania corporation

("Matthews"), Matthews, Cloverleaf Group, L.P., a Pennsylvania limited

partnership (the "Seller"), I.D.L. Incorporated, a Pennsylvania corporation

("iDL Inc."), Hugh Andrew, L.P., a Pennsylvania limited partnership ("Hugh Andrew, L.P."), Big Red Rooster, Inc., an Ohio corporation ("BRR Inc."),

Ralph W. Murray, Hugh A. Murray, Jeffrey H. Murray, Douglas G. Murray, Kristy

M. Holch, Cary M. Bohl, David G. Murray, C. Michael Dempe, 2000 Irrevocable

Trust for Family of Ralph W. Murray and 2000 Irrevocable Trust for Family of

Hugh A. Murray, (the "iDL Shareholders"), and Martin J. Beck, and Sweet Clover Group, Inc. (the "BRR Shareholders"). iDL Inc. and Hugh Andrew, L.P.

Clover Group, Inc. (the "BRR Shareholders"). 1DL Inc. and Hugh Andrew, L.I are sometimes referred to herein together as iDL.

WITNESSETH

WHEREAS, iDL Inc. and BRR Inc. have each contributed certain assets and liabilities to Seller;

WHEREAS, since the contribution of such assets, Seller has operated the business contributed by iDL Inc. and the business contributed by BRR Inc.

business contributed by iDL Inc. and the business contributed by BRR Inc.; and

WHEREAS, Buyer wishes to purchase or acquire from Seller, and Seller wishes to sell, assign and transfer to Buyer, all of the assets and properties held

by Seller (other than the Excluded Assets), and Buyer has agreed to assume certain Assumed Liabilities, all for the purchase price and upon the terms and subject to the conditions hereinafter set forth.

NOW, THEREFORE, in consideration of the mutual covenants, representations and warranties made herein, and of the mutual benefits to be derived hereby, and intending to be legally bound, the parties hereto agree as follows:

ARTICLE I

SALE AND PURCHASE OF THE ASSETS

1.1. Assets. Subject to and upon the terms and conditions set forth in this Agreement, Seller will sell, transfer, convey, assign and deliver to Buyer, and Buyer will purchase or acquire from Seller, other than the Excluded Assets, all right, title and interest of Seller in and to properties, assets and rights of every nature, kind and description, tangible and intangible, wherever located, whether real, personal or mixed, whether accrued,

contingent or otherwise and whether now existing or hereinafter acquired as the same may exist on the Closing Date (collectively, the "Assets"), including without limitation the following:

- (a) all machinery, equipment, furniture, furnishings, automobiles, trucks, vehicles, tools, dies, molds and parts and similar property (including, but not limited to, any of the foregoing purchased subject to any conditional sales or title retention agreement in favor of any other Person):
- (a) all inventories of raw materials, work in process, finished products, goods, spare parts, replacement and component parts, and office and other supplies wherever located (collectively, the "Inventories");
- (c) all rights in and to products sold or leased (including, but not limited to, products hereafter returned or repossessed and unpaid sellers' rights of rescission, replevin, reclamation and rights to stoppage in transit);
- (d) all rights (including but not limited to any and all Intellectual Property rights) in and to the products sold or leased and in and to any products or other Intellectual Property rights under research or development prior to or on the Closing Date;
- (e) all of the rights of Seller under all Contracts (as hereinafter defined), including, without limitation, any right to receive payment for products sold or services rendered, and to receive goods and services, pursuant to such agreements and to assert claims and take other rightful actions in respect of breaches, defaults and other violations of such contracts, arrangements, licenses, leases and other agreements and otherwise;
- (f) all credits, prepaid expenses, deferred charges, advance payments, security deposits and prepaid items;
- (g) all notes and accounts receivable held by Seller and all notes, bonds and other evidences of indebtedness of and rights to receive payments from any Person held by Seller;
- (h) all Intellectual Property and all rights thereunder or in respect thereof including, but not limited to, rights to sue for and remedies against past, present and future infringements thereof, and rights of priority and protection of interests therein under the laws of any jurisdiction worldwide and all tangible embodiments thereof (together with all Intellectual Property rights included in the other clauses of this Section 1.1, the "Intellectual Property Assets");
- (i) all books, records, manuals and other materials (in any form or medium), advertising matter, catalogues, price lists, correspondence, mailing lists, lists of customers, distribution lists, photographs, production data, sales and promotional materials and records, purchasing materials and records, personnel records, manufacturing and quality control records and procedures, blueprints, research and development files, records, data and laboratory books, Intellectual Property disclosures, media materials and plates, accounting records, sales order files and litigation files;
- (j) to the extent their transfer is permitted by law, all Governmental Approvals, including all applications therefor;
- (k) all Real Property and all licenses, permits, approvals and qualifications relating to any Real Property issued to Seller by any Governmental Authority;
- (1) all rights to causes of action, lawsuits, judgments, claims and demands of any nature available to or being pursued by Seller with respect to the ownership, use, function or value of any Asset, whether arising by way of counterclaim or otherwise; and
- (m) all guarantees, warranties, indemnities and similar rights in favor of Seller with respect to any Asset.
- 1.2. Excluded Assets.
- (a) Seller shall retain and not transfer to Buyer, and Buyer will not purchase or acquire from Seller the following assets (collectively, the

"Excluded Assets"):

- (i) Excess Cash and Cash Equivalents (as defined below) of Seller equal to the Calculated Amount; as used herein "Calculated Amount" shall mean \$4,115,000 less the following:
- (1) distributions to iDL Inc. shareholders for 2004 estimated income tax liabilities,
- (2) all expenses, incurred in 2004, including professional fees and costs, in connection with the formation of Seller,
- (3) all expenses, incurred in 2004, including professional fees and costs, associated with the proposed PNC financing with respect to Seller,
- (4) all expenses, incurred in 2004, including professional fees and costs, associated with the transactions contemplated by this Agreement and the Collateral Agreements, and
- (5) the \$1,000,000 intercompany loan made by iDL Inc. to Seller, which \$1,000,000 was subsequently loaned by Seller to BRR Inc. and then distributed by BRR Inc. to certain shareholders and employees.

Notwithstanding subsection 1.2(a)(i) above, Seller shall transfer to Buyer \$2,000,000 in cash and cash equivalents, and the cash and cash equivalents which are the Excluded Assets described in such subsection above shall only be any cash and cash equivalents in excess of such \$2,000,000 amount (such cash and cash equivalents which are in excess of such \$2,000,000 amount are referred to herein as the "Excess Cash and Cash Equivalents"). If Excess Cash and Cash Equivalents of Seller are not sufficient to equal the Calculated Amount, then certain trade accounts receivable to be agreed upon by Seller and Buyer (the "Retained Accounts Receivable") shall also be Excluded Assets, such that the Excess Cash and Cash Equivalents plus such Retained Accounts Receivable on the Closing Date shall equal the Calculated Amount. If the Excess Cash and Cash Equivalents of Seller exceed the

Calculated Amount, then such excess shall be retained by Buyer. A schedule of how such calculation shall be made under this Section 1.2(a)(i) is attached hereto as Schedule 1.2(a)(i). Such calculation shall be agreed to by Matthews and Seller within 20 days after the Closing Date.

(ii) the iDL building, land and related improvements in Plum Borough, Pennsylvania located at 535 Old Frankstown Road, Pittsburgh 15239 (the "Plum Facility"):

- (iii) the iDL building, land and related improvements in Murrysville, Pennsylvania located at 4425 West Penn Highway, Murrysville, 15668 (the "Murrysville Facility");
- (iv) two (2) notes receivable, each dated December 31, 2003, in the principal amounts of \$729,240 and \$857,148 made by the 2000 Irrevocable Trust for Family of Ralph W. Murray and the 2000 Irrevocable Trust for Family of Hugh A. Murray, respectively, related to the cash surrender assets of the life insurance policies of the principal officers of iDL Inc., namely Hugh A. Murray and Ralph W. Murray;
- (v) a shareholder note receivable from Michael Dempe in the amount of approximately \$986,000;
- (vi) an identified lot of finished goods inventory manufactured before December 31, 2003 that is a component of the reimaging program with iDL's customer BP having an approximate book value of \$1,800,000, if such inventory has not been shipped on the Closing Date (the "BP Inventory"), or any accounts receivable related to the BP Inventory if it has been shipped prior to the Closing Date; and
- (vii) iDL's equity interest in Goliath LLC, an Illinois limited liability company.
- (b) Seller shall retain any intercompany accounts receivable or payable resulting from transactions among iDL Inc., BRR Inc. and Seller, including the intercompany receivable from BRR Inc. related to the transaction described in Section 1.2(a)(i)(5) above.

ARTICLE II

THE CLOSING

- 2.1. Place and Date. The closing of the sale and purchase of the Assets from Seller (the "Closing") shall take place upon execution and delivery of this Agreement and all acts and deliveries provided by Article V (such date and time herein referred to as the "Closing Date").
- 2.2. Purchase Price. On the terms and subject to the conditions set forth in this Agreement, Buyer agrees to (a) pay or cause to be paid to Seller an aggregate of \$32,390,000, as such amount may be adjusted in accordance with Section 1.2(a)(i) (the "Adjusted Payment"), at the Closing, (b) make the Contingent Consideration Payments to Seller as described in Section 2.3, and (c) assume the certain Assumed Liabilities as provided in Section 2.5 (the amounts referred to subsection (a), (b) and (c) above being referred to

herein collectively as the "Purchase Price"). The portion of the Purchase Price described in subsection (a) above shall be payable at the Closing, by the wire transfer in immediately available funds to such bank account or accounts as per written instructions of the applicable party given to Buyer at least five days prior to the Closing as follows:

- (i) subject to Section 1.2(a)(i), to Seller in the amount of \$31,890,000; and
- (ii) to the Escrow Fund in the amount of \$500,000.

The "Escrow Fund" shall be maintained and administered pursuant to the Escrow Agreement executed and delivered as of the date hereof (the "Escrow Agreement").

- 2.3. Contingent Consideration. For the purposes of this Agreement, the following terms shall have the following meanings:
- (a) (i) "Operating Profit" shall mean the earnings of Buyer calculated in accordance with GAAP, excluding (x) any interest on indebtedness, other than interest on Temporary Working Capital Infusions (as described in subsection (b) below), which interest on Temporary Working Capital Infusions shall be included (i.e. deducted) when calculating Operating Profit, (y) federal and state income-based taxes, and (z) any adjustments made for purchase accounting purposes resulting from the transaction contemplated by this Agreement which have an impact on future reported Operating Profit, including depreciation and amortization of the re-valuation for purchase accounting purposes of tangible and intangible assets, and the cost of sales impact related to the re-valuation of inventory for purchase accounting purposes, in each case resulting from the transaction contemplated by this Agreement (but nevertheless including (and not excluding) all purchase accounting adjustments related to all future acquisitions). For the purposes of calculating Operating Profit, any direct administrative or financial services performed by Matthews or its Affiliates on behalf of Buyer (including without limitation accounts receivable, payroll, human resources, bill payment, benefits, retirement or pension plans, etc.) (the "Matthews Services") shall be charged as an expense. These Matthews Services will only be performed by Matthews or its Affiliates on behalf of Buyer if Matthews and Buyer agree that there exists the opportunity for cost savings to both Matthews and Buyer or if any of Matthews or its Affiliates (including Buyer) is legally required to do so. The calculation of Operating Profit shall not include corporate headquarters building costs, as currently captured in Matthews "Department 90" account, and executive management compensation or severance, public company reporting and related costs, as currently captured in the Matthews' "Department 93" account (collectively, "Corporate Overhead"). Although Corporate Overhead is not included in the calculation of Operating Profit, Corporate Overhead shall be allocated to Buyer (i) for purposes of creating segmented financial statements for public disclosure and internal intercompany financial statements and (ii) for purposes of calculating Free
- (ii) "Average Operating Profit" shall mean the mathematical average between the respective Operating Profits of any two applicable fiscal years of Buyer.
- (iii) "Initial Investment" shall mean the Adjusted Payment payable as described in Section 2.2(a), plus the actual total amount of Assumed Loans described in Section 2.5(i).
- (iv) "Free Cash Flow" shall mean, for any period, the cash flow of Buyer as determined in accordance with GAAP, consistently applied, based upon the Buyer's net income for such period also determined in accordance with GAAP, consistently applied, plus for such period (A) the Buyer's allocated share of Corporate Overhead, (B) the Buyer's allocated share of the cost of any Matthews Services which were not used by Buyer, (C) the amount of any Temporary Working Capital Infusions converted to Subsequent Investment Cost incurred by Buyer during such period, and (D) solely for the first fiscal quarter in which Free Cash Flow is calculated, the total amount of the Assumed Loans. An example of a Free Cash Flow calculation is attached hereto as Schedule 2.3(a)(iv). The line items used on the Schedule are examples of line items in connection with the Free Cash Flow calculation, but the dollar amounts used in the example are fictitious.
- (v) "Accumulated Free Cash Flow" shall mean the aggregate (positive and negative) of Free Cash Flows for the period from the effective date of the Closing Net Assets Statement through the date Accumulated Free Cash Flow is being measured.
- (vi) "Initial Acquisition Cost" shall mean the Initial Investment plus an amount equal to 15% interest on the Initial Investment, compounded annually taking into account the number of annual periods or fraction thereof from the Closing Date until the end of an applicable period.

(vii) "Subsequent Investment Cost" shall mean any Cash Infusions (as defined in Section 2.3(b)) provided to Buyer by Matthews or its Affiliates plus (with respect to each Cash Infusion) an amount equal to 15% interest on the Cash Infusion, compounded annually taking into account the number of annual periods or fraction thereof since the date a Temporary Working Capital Infusion was converted to a Cash Infusion until the end of an applicable period.

(viii) "Outstanding Temporary Working Capital Infusions" shall mean the total outstanding amount of the Temporary Working Capital Infusions not repaid by Buyer to Matthews or its Affiliates as of the end of any Fiscal Quarter. (ix) "2006 Contingent Acquisition Cost" shall mean the 2006 Contingent Amount (as defined in Section 2.3(c)(i)) plus an amount equal to 15% interest on the 2006 Contingent Amount, compounded annually taking into account the number of annual periods or fraction thereof since each of the two payments have been made to Seller with respect to the 2006 Contingent Amount, until the end of an applicable period.

(x) "Fiscal Quarter" shall have the meaning set forth in Section 2.3(b)(i). (b) (i) Buyer shall be entitled to use its Free Cash Flow to fund its operations, acquisitions and other capital needs. To the extent Buyer requires temporary working capital cash or cash funding infusions beyond its Free Cash Flow (the "Temporary Working Capital Infusions") from Matthews or its Affiliates for any of the reasons set forth in the preceding sentence, or

to the extent that at the end of any fiscal quarter ending December 31, March 31, June 30 or September 30 (a "Fiscal Quarter"), Accumulated Free Cash Flow is calculated to be a negative number (a "Shortfall"), Matthews or its Affiliates will be deemed to have made an advance of funds for such Temporary Working Capital Infusions to Buyer, in an amount requested by Buyer or to make up any Shortfall. Temporary Working Capital Infusions shall bear simple interest at a rate equal to the Prime Rate on the date at the end of a Fiscal Quarter such Temporary Working Capital Infusion is made, as the case may be, plus 2%. The interest expense of Buyer for any Temporary Working Capital Infusions will be charged against Buyer Operating Profit after any such Temporary Working Capital Infusions are made, for purposes of calculating Operating Profit for this Agreement, until such time that the Temporary Working Capital Infusions amount is repaid by Buyer to Matthews or its Affiliates or until such Temporary Working Capital Infusions are converted into a Cash Infusion (as described below).

- (ii) Matthews shall prepare a statement of Free Cash Flow for each Fiscal Quarter, and the then current amount of Temporary Working Capital Infusions shall be redetermined based upon such statement and shall be deemed to remain outstanding at least until the end of the next Fiscal Quarter, and thereafter if so determined based upon the next Free Cash Flow Statement. By way of example, if Accumulated Free Cash Flow increases as between two successive Fiscal Quarters, then the amount of Outstanding Temporary Working Capital Infusions, as of the end of the second such Fiscal Quarter, shall be deemed to decrease by such amount of increase in Accumulated Free Cash Flow, and if Accumulated Free Cash Flow decreases as between two successive Fiscal Quarters, then the amount of Outstanding Temporary Working Capital Infusions as of the end of the second Fiscal Quarter, shall be deemed to increase by such amount of decrease in Accumulated Free Cash Flow, and in either case the then Outstanding Temporary Working Capital Infusions (if any) shall bear interest at the then applicable Prime Rate as of the date at the end of such second Fiscal Quarter, plus 2%.
- (iii) Temporary Working Capital Infusions which are not repaid within 18 months of the date made shall be deemed to be cash infusions (the "Cash Infusions"), which are then subject to a 15% cost of capital as described in the definition of "Subsequent Investment Cost." Conversely, the 15% cost of capital shall not be applied on Temporary Working Capital Infusions until they have been outstanding for 18 months and have therefore been converted to a Cash Infusion. By way of example, if a Temporary Working Capital Infusion of \$100,000 is made as of December 31, 2004, and as of the end of each successive Fiscal Quarter through June 30, 2006, the Outstanding Temporary Working Capital Infusions are never below \$20,000, then on June 30, 2006, \$20,000 shall be converted from an Outstanding Temporary Working Capital Infusion into a Cash Infusion.
- (c) As additional consideration with respect to the Purchase Price, Seller shall be eligible for contingent consideration payments (the "Contingent Consideration Payments") based upon the Operating Profit of Buyer during the six-year period commencing October 1, 2004 and ending on September 30, 2010 (the "Contingent Payment Period"). Such Contingent Consideration Payments, if any, shall be made by Buyer to Seller in accordance with written

instructions as to place of delivery to be provided by Seller to Matthews before the date of any such payments. Such Contingent Consideration Payments

shall be further distributed to the partners of Seller in accordance with the provisions of Schedule 2.3(c). Such distribution procedures shall not be modified without the consent of Matthews. Based on a calculated statement to determine Contingent Consideration Payments with respect to Buyer for the applicable periods, which will be prepared and certified by Buyer's Auditors (the "Calculated Statement"), Seller shall be entitled to the following Contingent Consideration Payments, if any:

- (i) In the event that six (6) multiplied by the Average Operating Profit for the two fiscal years ended September 30, 2005 and September 30, 2006 is in excess of the sum of: (A) the Initial Acquisition Cost calculated through September 30, 2006; (B) any Subsequent Investment Cost calculated through September 30, 2006; and (C) any Outstanding Temporary Working Capital Infusions at September 30, 2006, Matthews shall contribute to Buyer and Buyer shall pay fifty percent (50%) of such excess to Seller (the "2006 Contingent Amount") in two equal installments, the first of which shall be payable on January 2, 2007 and the second of which, together with simple interest thereon from January 2, 2007 at the Prime Rate, shall be payable on January 2, 2008.
- (ii) In the event that six (6) multiplied by the Average Operating Profit for the two fiscal years ended September 30, 2007 and September 30, 2008 is in excess of the sum of: (A) the Initial Acquisition Cost calculated through September 30, 2008; (B) any 2006 Contingent Acquisition Cost calculated through September 30, 2008; (C) any Subsequent Investment Cost calculated through September 30, 2008; and (D) any Outstanding Temporary Working Capital Infusions at September 30, 2008, Matthews shall contribute to Buyer and Buyer shall pay fifty percent (50%) of such excess to Seller (the "2008 Contingent Amount") in two equal installments, the first of which shall be payable on January 2, 2009 and the second of which, together with simple interest thereon from January 2, 2009 at the Prime Rate, shall be payable on January 2, 2010; and
- (iii) In the event that six (6) multiplied by the Average Operating Profit for the two fiscal years ended September 30, 2009 and September 30, 2010 is in excess of the sum of: (A) the Initial Acquisition Cost calculated through September 30, 2010; (B) any 2006 Contingent Acquisition Cost calculated through September 30, 2010; (C) any 2008 Contingent Amount plus an amount equal to 15% interest on the 2008 Contingent Amount, compounded annually taking into effect the number of annual periods or fraction thereof since each of the two payments have been made to Seller with respect to the 2008 Contingent Amount, calculated through September 30, 2010; (D) any Subsequent Investment Cost calculated through September 30, 2010; and (E) any Outstanding Temporary Working Capital Infusions at September 30, 2010, Matthews shall contribute to Buyer and Buyer shall pay fifty percent (50%) of such excess to Seller in two equal installments, the first of which shall be payable on January 2, 2011 and the second of which, together with simple interest thereon from January 2, 2011 at the Prime Rate, shall be payable on January 2, 2012.
- (iv) There shall no minimum or maximum Contingent Consideration Payments that may be earned.
- 2.4 Closing Date Net Asset Statement. As promptly as practicable after the Closing Date (but in no event later than 60 days after the Closing Date), the Buyer will deliver to Seller a statement of the net assets for each of the respective net assets transferred by iDL Inc. and BRR Inc. as of the Closing Date (the "Closing Net Assets Statements"). The Closing Net Assets Statements will be prepared and certified on an accrual basis by Buyer's Auditor in accordance with GAAP. The fees and expenses of Buyer's Auditor in preparing and certifying the Closing Net Assets Statements shall be shared equally between Seller and Matthews.
- 2.3. Assumption of Liabilities. Subject to the terms and conditions set forth herein, at the Closing Buyer shall assume and agree to pay, honor and discharge when due all of the following liabilities relating to the Assets and existing at or arising on or after the Closing Date (collectively, the "Assumed Liabilities"):
- (a) any and all Working Capital Liabilities (as hereinafter defined) of Seller that are (i) reflected on the Balance Sheets or (ii) incurred after the date of the Balance Sheets in the ordinary course of business consistent with prior practice and in accordance with the terms of this Agreement. For purposes hereof, "Working Capital Liabilities" shall include trade accounts payable, accrued compensation and benefits, accrued vacation, other accrued

expenses, all accrued and unpaid interest on the Assumed Loans (as defined below), and other current liabilities;

- (b) any and all liabilities, obligations and commitments accruing and arising after the Closing Date out of the Contracts set forth on Schedule 3.1.12(a), but not including any obligation or liability for any breach thereof occurring prior to the Closing Date; and
- (c) liabilities in respect of Transferred Employees and employee benefit plans to the extent specifically assumed by Buyer pursuant to Article VI;
- (d) the equipment financing loan between iDL and the Commonwealth of Pennsylvania Department of Community and Economic Development Machinery and Equipment Loan Fund ("MELF") dated October 14, 1998;
- (e) the Loan between iDL and PNC Bank N.A. dated as of May 20, 2002;
- (f) the equipment financing loan between BRR and Bank One dated September 17, 2002, as amended December 23, 2002;
- (g) the BRR loan from Fifth Third Bank (Chicago), dated October 1, 2003 in the original principal amount of \$400,000;
- (h) the BRR loan from Fifth Third Bank (Central Ohio), dated July 1, 2003 in the original principal amount of \$200,000:
- (i) the Sweet Clover Group, Inc. loan to BRR dated October 29, 2003 in the amount of \$50,000; and
- (j) any other recorded interest-bearing indebtedness (including capitalized lease obligations) as of the Closing Date (items (d), (e), (f), (g), (h) and
- (i) being referred to herein as the "Assumed Loans"). The actual amount of the Assumed Loans on the Closing Date is set forth herein on Schedule 2.5(j). 2.4. Excluded Liabilities. Notwithstanding the provisions of Section 2.5 or any other provision hereof or any schedule or exhibit hereto and regardless of any disclosure to Buyer, Buyer shall not assume any liabilities, obligations or commitments of Seller relating to or arising out of the operation of Seller's business or the ownership of the Assets prior to the Closing other than the Assumed Liabilities (the "Excluded Liabilities") including, without limitation, the following:
- (i) any intercompany accounts payable resulting from transactions among Seller, iDL Inc. and BRR Inc., including the intercompany payable from Seller to iDL Inc. related to the transaction described in Section 1.2(a)(i)(5);
- (ii) income tax liabilities of iDL, BRR Inc., or Seller or their shareholders or partners respectively;
- (iii) any liabilities associated with the formation of Seller, including without limitation the PNC loan to Seller, if any, as of the Closing Date;
- (iv) any liabilities not contained on the Balance Sheets; and
- (v) any liabilities arising under any applicable Environmental Laws, relating in any way to the Seller's operation of the business or to any environmental conditions on, at, under or from the Real Property, the Assets or other properties or assets owned, leased or used by Seller existing at or prior to the Closing Date.
- 2.7. Allocation of Purchase Price.
- (a) The parties agree to allocate the Purchase Price among the Assets in accordance with an allocation schedule reasonably acceptable to Seller which shall be prepared by Buyer. Buyer intends to obtain an appraisal with respect to the Assets. Such allocation schedule shall be prepared in accordance with section 1060 of the Code and in accordance with such appraisal.
- (b) In connection with the determination of the foregoing allocation schedule, the parties shall cooperate with each other and provide such information as any of them shall reasonably request. The parties will each report the federal, state and local and other Tax consequences of the purchase and sale contemplated hereby (including the filing of Internal Revenue Service Form 8594) in a manner consistent with such allocation schedule.

ARTICLE III

REPRESENTATIONS AND WARRANTIES

3.1. Representations and Warranties of Seller. The parties acknowledge that the Assets are contemporaneously being transferred from iDL Inc. and BRR Inc. to Seller, and then to Buyer. In view of the foregoing, and for the avoidance of doubt, the use of the term "Seller" in this Article III shall include each of the corporate entities iDL Inc. and BRR Inc., as the case may be, and shall refer to the Assets as held by Seller and as held by each of IDL Inc. and BRR Inc., as the case may be. Each of Seller and the BRR Group jointly and severally represent and warrant to Buyer with respect to the Assets which were owned by BRR Inc., and each of Seller and the iDL Group

jointly and severally represent to Buyer with respect to the Assets which were owned by iDL, as follows:

3.1.1. Authorization, etc. Each of the Seller, iDL Inc., Hugh Andrew, L.P., BRR Inc., the iDL Shareholders and BRR Shareholders has the requisite applicable power and authority to execute and deliver this Agreement and each of the Collateral Agreements to which it will be a party, to perform fully its obligations thereunder, and to consummate the transactions contemplated hereby and thereby. The execution and delivery by such parties of this Agreement and the Collateral Agreements to which it is a party, and the consummation of the transactions contemplated hereby and thereby, have been duly authorized by all requisite action of such parties. Such parties have duly executed and delivered this Agreement and the Collateral Agreements to which it is a party. This Agreement and each of the Collateral Agreements is a legal, valid and binding obligations of such parties, enforceable against such parties in accordance with their respective terms.

3.1.2. Corporate Status.

- (a) Seller is a limited partnership duly organized, validly existing and in good standing under the laws of the Commonwealth of Pennsylvania, with full partnership power and authority to carry on its business and to own or lease and to operate its properties as and in the places where such business is conducted and such properties are owned, leased or operated.
- (b) Seller is duly qualified or licensed to do business and is in good standing in each of the jurisdictions specified opposite its name in Schedule 3.1.2(b), which are the only jurisdictions in which the operation of its business or the character of the properties owned, leased or operated by it makes such qualification or licensing necessary unless the failure to so qualify would not result in a Material Adverse Effect.
- (c) Seller has delivered to Buyer complete and correct copies of its certificate of limited partnership and partnership agreement or other organizational documents, in each case, as amended and in effect on the date hereof. Seller is not in violation of any of the provisions of its certificate of limited partnership or other organizational documents. 3.1.3. No Conflicts, etc. The execution, delivery and performance by each of the Seller, iDL Inc., Hugh Andrew, L.P., BRR Inc., the iDL Shareholders and the BRR Shareholders of this Agreement and each of the Collateral Agreements to which it is a party, and the consummation of the transactions contemplated hereby and thereby, do not and will not conflict with or result in a violation of or a default under (with or without the giving of notice or the lapse of time or both) (i) any Applicable Law applicable to such party or any of the properties or assets of such party (including but not limited to the Assets), (ii) the certificate of formation or operating agreement or other organizational documents of such party or (iii) except as set forth in Schedule 3.1.3, any Contract or other material contract, agreement or other instrument to which such party is a party or by which such party or any of its properties or assets, including but not limited to the Assets, may be bound or affected. Except as specified in Schedule 3.1.3, no Governmental Approval or other Consent is required to be obtained or made by such party in connection with the execution and delivery of this Agreement and the Collateral Agreements or the consummation of the

transactions contemplated hereby or thereby.

3.1.4. Financial Statements. The Buyer has been delivered (a) (i) audited consolidated financial statements of iDL as at and for the periods ended January 3, 2004 (the "Audited Balance Sheet Date"), together with a report thereon by iDL's accountants (the "Audited Financial Statements"), (ii) unaudited consolidated financial statements of iDL as at and for the period ended March 31, 2004 (the "Unaudited Quarterly iDL Financial Statements"), (iii) unaudited consolidated financial statements of iDL as at and for the monthly periods ended April 30 and May 31, 2004 (the "Monthly iDL Unaudited Financial Statements"), including in each of clauses (i), (ii) and (iii) a balance sheet, statements of income and retained earnings and in clause (i) a statement of cash flows and (b) (i) unaudited consolidated financial statements of BRR as at and for the periods ended December 31, 2003 (the "Unaudited Balance Sheet Date") (the "Unaudited Annual Financial Statements"), (ii) unaudited consolidated financial statements of BRR as at and for the period ended March 31, 2004 (the "Unaudited Quarterly BRR Financial Statements"), (iii) unaudited consolidated financial statements of BRR as at and for the monthly periods ended April 30 and May 31, 2004 (the "Monthly BRR Unaudited Financial Statements"), including in each of clauses (i), (ii) and (iii) a balance sheet, statements of income and retained earnings and in clause (i) a statement of cash flows (the Audited Financial Statements, Unaudited Annual Financial Statements, the Unaudited Ouarterly iDL Financial Statements, the Unaudited Quarterly BRR Financial Statements, the Monthly iDL Unaudited Financial Statements and the Monthly BRR Unaudited Financial Statements are referred to herein collectively the "Financial Statements"). The Audited Financial Statements are complete and correct in all material respects and have been prepared in accordance with GAAP applied throughout the periods indicated. The Unaudited Quarterly iDL Financial Statements and the Monthly iDL Unaudited Financial Statements have been prepared in all material respects on a basis consistent with the Audited Financial Statements, except that the Unaudited Quarterly iDL Financial Statements and the Monthly iDL Unaudited Financial Statements do not contain notes and may be subject to normal audit adjustments and, in the case of the Monthly iDL Unaudited Financial Statements, normal annual adjustments. The Unaudited Annual Financial Statements are complete and correct in all material respects and have been prepared in accordance with GAAP applied throughout the periods indicated. The Unaudited Quarterly BRR Financial Statements and the Monthly BRR Unaudited Financial Statements have been prepared in all material respects on a basis consistent with the Unaudited Annual Financial Statements, except that the Unaudited Quarterly BRR Financial Statements and the Monthly BRR Unaudited Financial Statements do not contain notes and may be subject to normal audit adjustments and, in the case of the Monthly BRR Unaudited Financial Statements, normal annual adjustments. The balance sheets included in the Financial Statements do not include any material assets or liabilities not intended to constitute a part of the Assets after giving effect to the transactions contemplated hereby, and present fairly the financial condition of Seller as at their respective dates. The statements of income and retained earnings and statements of cash flows included in the Financial Statements do not reflect the operations of any entity or business not intended to constitute a part of Seller's business after giving effect to all such transactions, reflect all costs that historically have been incurred by Seller (other than the Excluded Liabilities) and present fairly the results of operations and cash flows of Seller for the periods indicated.

3.1.5. Absence of Undisclosed Liabilities. Seller has no liabilities or obligations of any nature, whether known or unknown, absolute, accrued, contingent or otherwise and whether due or to become due, except as and to the extent reflected or reserved against in the Balance Sheets (excluding the notes thereto) and for liabilities and obligations that (i) were incurred after the date of the Balance Sheets in the ordinary course of business consistent with prior practice and (ii) individually and in the aggregate are not material to the Seller and have not had or resulted in, and will not have or result in, a Material Adverse Effect. None of the Seller's employees are now or will by the passage of time hereinafter become entitled to receive any vacation time, vacation pay or severance pay attributable to services rendered prior to such date except as disclosed on the Balance Sheets (excluding the notes thereto) or otherwise described on Schedule 3.1.5. 3.1.6. Taxes.

(a) Seller has (or by the Closing will have) duly and timely filed all Tax Returns with respect to Taxes to be filed on or before the Closing Date ("Tax Returns"). Except for Taxes set forth on Schedule 3.1.6(a), which are being contested in good faith and by appropriate proceedings, the following Taxes have (or by the Closing Date will have) been duly and timely paid: (i) all Taxes shown to be due on the Returns, (ii) all deficiencies and assessments of Taxes of which notice has (or by the Closing Date will have) been received by Seller that are or may become payable by Buyer chargeable as a lien upon the Assets and (iii) all other Taxes due and payable on or before the Closing Date for which neither filing of Tax Returns nor notice of deficiency or assessment is required, if Seller is or reasonably should be (or by the Closing Date will be or reasonably should be) aware that are or may become payable by Buyer or chargeable as a lien upon the Assets. All Taxes required to be withheld by or on behalf of Seller in connection with amounts paid or owing to any employee, independent contractor, creditor or other party ("Withholding Taxes") have been withheld, and such withheld taxes have either been duly and timely paid to the proper Governmental Authorities or set aside in accounts for such purpose.

(b) Except as set forth on Schedule 3.1.6(b), no agreement or other document extending, or having the effect of extending, the period of assessment or collection of any Taxes or Withholding Taxes, and no power of

attorney with respect to any such Taxes, has been filed with the IRS or any other Governmental Authority.

(c) Except as set forth on Schedule 3.1.6(c), (i) there are no Taxes or Withholding Taxes asserted in writing by any Governmental Authority to be due and (ii) no issue has been raised in writing by any Governmental Authority in the course of any audit with respect to Taxes or Withholding Taxes. Except as set forth on Schedule 3.1.6(c), no Taxes and no Withholding

Taxes are currently under audit by any Governmental Authority. Except as set forth on Schedule 3.1.6(c), neither the IRS nor any other Governmental Authority is now asserting or, to the best knowledge of Seller, threatening to assert against Seller any deficiency or claim for additional Taxes or any adjustment of paid Taxes that would, if paid by Buyer, have a Material Adverse Effect, and there is no reasonable basis for any such assertion of which Seller is or reasonably should be aware.

- (d) Buyer will not be required to deduct and withhold any amount pursuant to section 1445(a) of the Code upon the transfer of the Assets to Buyer.
- (e) Except as set forth on Schedule 3.1.6(e), there is no litigation or administrative appeal pending or, to the knowledge of Seller, threatened against or relating to Seller in connection with paid Taxes.
- 3.1.7. Absence of Changes. Except as set forth in Schedule 3.1.7, since May 31, 2004, Seller has conducted its business only in the ordinary course consistent with prior practice and has not, on behalf of, in connection with or relating to the business or the Assets:
- (a) suffered any Material Adverse Effect;
- (b) incurred any obligation or liability, absolute, accrued, contingent or otherwise, whether due or to become due, except current liabilities for trade or business obligations incurred in connection with the purchase of goods or services in the ordinary course of business consistent with prior practice, none of which liabilities, in any case or in the aggregate, could have a Material Adverse Effect;
- (c) discharged or satisfied any Lien other than those then required to be discharged or satisfied, or paid any obligation or liability, absolute, accrued, contingent or otherwise, whether due or to become due, other than current liabilities shown on the Balance Sheets and current liabilities incurred since the date thereof in the ordinary course of business consistent with prior practice;
- (d) mortgaged, pledged or subjected to Lien, any property, business or assets, tangible or intangible;
- (e) sold, transferred, leased to others or otherwise disposed of any of the Assets, except for inventory sold in the ordinary course of business, or cancelled or compromised any debt or claim, or waived or released any right of substantial value,
- (f) received any notice of termination of any contract, lease or other agreement or suffered any damage, destruction or loss (whether or not covered by insurance) which, in any case or in the aggregate, has had a Material Adverse Effect;
- (g) transferred or granted any rights under, or entered into any settlement regarding the breach or infringement of, any Intellectual Property, or modified any existing rights with respect thereto;
- (h) made any change in the rate of compensation, commission, bonus or other direct or indirect remuneration payable, or paid or agreed or orally promised to pay, conditionally or otherwise, any bonus, incentive, retention or other compensation, retirement, welfare, fringe or severance benefit or vacation pay, to or in respect of any shareholder, director, officer, employee, salesman, distributor or agent of Seller;
- (i) encountered any labor union organizing activity, had any actual or threatened employee strikes, work stoppages, slowdowns or lockouts, or had any material change in its relations with its employees, agents, customers or suppliers:
- (j) failed to replenish inventories and supplies in a normal and customary manner consistent with its prior practice and prudent business practices prevailing in the industry, or made any purchase commitment in excess of the normal, ordinary and usual requirements of its business or at any price in excess of the then current market price or upon terms and conditions more onerous than those usual and customary in the industry, or made any change in its selling, pricing, advertising or personnel practices inconsistent with its prior practice and prudent business practices prevailing in the industry;
- (k) made any capital expenditures or capital additions or improvements in excess of an aggregate of \$50,000;
- (1) instituted, settled or agreed to settle any litigation, action or proceeding before any court or governmental body other than in the ordinary course of business consistent with past practices but not in any case involving amounts in excess of \$50,000;
- (m) entered into any transaction, contract or commitment other than in the ordinary course of business or paid or agreed to pay any legal, accounting, brokerage, finder's fee, Taxes or other expenses in connection with, or incurred any severance pay obligations by reason of, this Agreement or the transactions contemplated hereby;

- (n) made any dividend, equity redemption or other distribution other than dividends for 2003 taxes made prior to April 15, 2004, and dividends for 2004 estimated tax payments, in each case as previously disclosed to Matthews; or
- (o) taken any action or omitted to take any action that would result in the occurrence of any of the foregoing.
- 3.1.8. Litigation. Except as set forth on Schedule 3.1.8, there is no action, claim, demand, suit, proceeding, arbitration, grievance, citation, summons, subpoena, inquiry or investigation of any nature, civil, criminal, regulatory or otherwise, in law or in equity, pending or threatened against or relating to Seller or against or relating to the transactions contemplated by this Agreement, and Seller does not know and has no reason to be aware of any basis for the same. Except as set forth in such Schedule 3.1.8, no citations, fines or penalties have been asserted against Seller under any Environmental Law or any foreign, federal, state or local law relating to occupational health or safety.
- 3.1.9. Compliance with Laws; Governmental Approvals and Consents; Governmental Contracts.
- (a) Except as disclosed in Schedule 3.1.9(a), Seller has complied in all material respects with all Applicable Laws, and Seller has not received any notice alleging any such conflict, violation, breach or default.
- (b) Schedule 3.1.9(b) sets forth all Governmental Approvals and other Consents necessary for, or otherwise material to, the conduct of its business. Except as set forth in Schedule 3.1.9(b), all such Governmental Approvals and Consents have been duly obtained and are in full force and effect, and Seller is in compliance with each of such Governmental Approvals and Consents held by it.
- (c) Schedule 3.1.9(c) sets forth all Contracts with any Governmental Authority.
- (d) There are no proposed laws, rules, regulations, ordinances, orders, judgments, decrees, governmental takings, condemnations or other proceedings which would be applicable to the business, operations or properties of Seller and which might adversely affect the properties, assets, liabilities, operations or prospects of Seller either before or after the Closing Date.
- 3.1.10. Operation of the Business. Except as set forth in Schedule 3.1.10, the Assets (a) constitute all of the assets, tangible and intangible, necessary to operate Seller's business in the manner presently operated by Seller and (b) include all the operating assets of Seller.
- 3.1.11. Assets. Except as disclosed in Schedule 3.1.11, Seller has good title to all the Assets free and clear of any and all Liens other than Permitted Liens. Except for Excluded Assets, there are no assets or properties used in the operation of Seller's business and owned by any Person other than Seller that will not be leased or licensed to Buyer under valid, current leases or license arrangements. The Assets are in all material respects adequate for the purposes for which such assets are currently used or are held for use, and are in reasonably good repair and operating condition (subject to normal wear and tear) and, to the knowledge of Seller, there are no facts or conditions affecting the Assets which could, individually or in the aggregate, interfere in any material respect with the use, occupancy or operation thereof as currently used, occupied or operated, or their adequacy for such use.

3.1.12. Contracts.

- (a) Schedule 3.1.12(a) contains a complete and correct list of all agreements, contracts, commitments and other instruments and arrangements (whether written or oral) of the types described below (x) under which Seller has or may acquire any rights or benefits (y) under which Seller has or may become subject to any obligation or liability; or (z) by which Seller or any of the Assets is or may be bound (the "Contracts"):
- (i) leases, licenses, permits, franchises, insurance policies, Governmental Approvals and other contracts concerning or relating to the Real Property:
- (ii) employment, consulting, agency, collective bargaining or other similar contracts, agreements, and other instruments and arrangements relating to or for the benefit of current, future or former employees, officers, directors, sales representatives, distributors, dealers, agents, independent contractors or consultants;
- (iii) loan agreements, indentures, letters of credit, mortgages, security agreements, pledge agreements, deeds of trust, bonds, notes, guarantees, and other agreements and instruments relating to the borrowing of

money or obtaining of or extension of credit;

- (iv) licenses, licensing arrangements and other contracts providing in whole or in part for the 'use of, or limiting the use of, any Intellectual Property;
- (v) brokerage or finder's agreements;
- (vi) joint venture, partnership and similar contracts involving a sharing of profits or expenses (including but not limited to joint research and development and joint marketing contracts);
- (vii) asset purchase agreements and other acquisition or divestiture agreements, including but not limited to any agreements relating to the sale, lease or disposal of any Assets (other than sales of inventory in the ordinary course of business) or involving continuing indemnity or other obligations;
- (viii) orders and other contracts for the purchase or sale of materials, supplies, products or services, each of which involves aggregate payments in excess of \$50,000;
- (ix) contracts with respect to which the aggregate amount that could reasonably expected to be paid or received thereunder in the future exceeds \$50,000 per annum or \$500,000 in the aggregate;
- (x) sales agency, manufacturer's representative, marketing or distributorship agreements;
- (xi) contracts, agreements or arrangements with respect to the representation of Seller in foreign countries;
- (xii) master lease agreements providing for the leasing of personal property used by Seller;
- (xiii) contracts, agreements or commitments with any employee, director, officer, stockholder or Affiliate of Seller; and
- (xiv) any other contracts, agreements or commitments that are material to the Seller.
- (b) Seller has delivered to Buyer complete and correct copies of all written Contracts, together with all amendments thereto, and accurate descriptions of all material terms of all oral Contracts, set forth or required to be set forth in Schedule 3.1.12(a).
- (c) All Contracts are in full force and effect and enforceable against each party thereto. There does not exist under any Contract any event of default or event or condition that, after notice or lapse of time or both, would constitute a violation, breach or event of default thereunder on the part of Seller or, to the best knowledge of Seller, any other party thereto except as set forth in Schedule 3.1.12(c) and except for such events or conditions that, individually and in the aggregate, (i) has not had or resulted in, and will not have or result in, a Material Adverse Effect and (ii) has not and will not materially impair the ability of Seller to perform its obligations under the this Agreements and under the Collateral Agreements. Except as set forth in Schedule 3.1.12(c), no consent of any third party is required under any Contract as a result of or in connection with, and the enforceability of any Contract will not be affected in any manner by, the execution, delivery and performance of this Agreement, or any of the Collateral Agreements or the consummation of the transactions contemplated hereby or thereby.
- 3.1.13. Territorial Restrictions. Seller is not restricted by any written agreement or understanding with any other Person from carrying on its business anywhere in the world. Buyer, solely as a result of its purchase of the Assets from Seller pursuant hereto and the assumption of the Assumed Liabilities, will not thereby become restricted in carrying on any business anywhere in the world.
- 3.1.14. Inventories. All Inventories are of good, usable and merchantable quality in all material respects and, except as set forth on Schedule 3.1.14, do not include obsolete or discontinued items. Except as set forth on Schedule 3.1.14, (a) all Inventories are of such quality as to meet the quality control standards of Seller and any applicable governmental quality control standards, (b) all Inventories that are finished goods are saleable as current inventories at the current prices thereof in the ordinary course of business, (c) all Inventories are recorded on the books of Seller at the lower of cost or market value determined in accordance with GAAP and (d) no write-down in inventory has been made or should have been made pursuant to GAAP during the past two years.
- 3.1.15. Customers. Schedule 3.1.15 sets forth (a) the names and addresses of all customers of Seller that ordered goods and services from Seller with an aggregate value for each such customer of \$50,000 or more

during the twelve-month period ended May 31, 2004 and (b) the amount for which each such customer was invoiced during such period. Seller has not received any notice or has any reason to believe that any significant customer of Seller (i) has ceased, or will cease, to use the products, goods or services of Seller, (ii) has substantially reduced or will substantially reduce, the use of products, goods or services of Seller or (iii) has sought, or is seeking, to reduce the price it will pay for products, goods or services of Seller, including in each case after the consummation of the transactions contemplated hereby. To the best knowledge of Seller, no customer described in clause (a) of the first sentence of this Section has otherwise threatened to take any action described in the preceding sentence as a result of the consummation of the transactions contemplated this Agreement and the Collateral Agreements.

3.1.16. Suppliers; Raw Materials. Schedule 3.1.16 sets forth (a) the names and addresses of all suppliers (including without limitation Seller and any Affiliates thereof) from which Seller ordered raw materials, supplies, merchandise and other goods and services with an aggregate purchase price for each such supplier of \$50,000 or more during the twelve-month period ended May 31, 2004 and (b) the amount for which each such supplier invoiced Seller during such period. Seller has not received any notice or has any reason to believe that there has been any material adverse change in the price of such raw materials, supplies, merchandise or other goods or services, or that any such supplier will not sell raw materials, supplies, merchandise and other goods to Buyer at any time after the Closing Date on terms and conditions similar to those used in its current sales to Seller, subject to general and customary price increases. To the best knowledge of Seller, no supplier of Seller described in clause (a) of the first sentence of this Section has otherwise threatened to take any action described in the preceding sentence as a result of the consummation of the transactions contemplated by this Agreement, and the Collateral Agreements.

3.1.17. Product Warranties. Except as set forth in Schedule 3.1.17 and for warranties under Applicable Law, (a) there are no warranties express or implied, written or oral, with respect to the products of Seller and (b) there are no pending or threatened claims with respect to any such warranty, and Seller has no liability with respect to any such warranty, whether known or unknown, absolute, accrued, contingent or otherwise and whether due or to become due

3.1.18. Absence of Certain Business Practices. Neither Seller, any officer, employee or agent of Seller, nor any other person acting on their behalf, has, directly or indirectly, within the past five years given or agreed to give any gift or similar benefit to any customer, supplier, governmental employee or other person who is or may be in a position to help or hinder Seller (i) which subjected or might have subjected Seller to any damage or penalty in any civil, criminal or governmental litigation or proceeding, (ii) which if not given in the past, might have had a Material Adverse Effect, (iii) which if not continued in the future, might have a Material Adverse Effect or subject Seller to suit or penalty in any private or governmental litigation or proceeding, (iv) for any of the purposes

described in Section 162(c) of the Code or (y) for the purpose of establishing or maintaining any concealed fund or concealed bank account. 3.1.19. Intellectual Property. Schedule 3.1.19(a) sets forth a complete and correct list of all material Intellectual Property that is owned by Seller (the "Owned Intellectual Property"). The Owned Intellectual Property constitutes all material Intellectual Property used or held for use in connection with, necessary for the conduct of, or otherwise material to Seller's business, except as set forth on Schedule 3.1.19(b). Immediately after the Closing, Buyer will have the right to use all Intellectual Property described on Schedule 3.1.19(a) and will own all Owned Intellectual Property, free from any Liens (other than Permitted Liens). Schedule 3.1.19(c) sets forth a complete and correct list of all material written or oral licenses and arrangements, (i) pursuant to which the use by any Person of Intellectual Property is permitted by Seller and (ii) pursuant to which the use by Seller of Intellectual Property is permitted by any Person (collectively, the "Intellectual Property Licenses"). All Intellectual Property Licenses are in full force and effect in accordance with their terms, and are free and clear of any Liens (other than Permitted Liens). Seller is not in default under any Intellectual Property License, and no such default is currently threatened. To the knowledge of Seller, the conduct of Seller's business does not infringe the rights of any third party in respect of any Intellectual Property, except as set forth on Schedule 3.1.19(d). To the knowledge of Seller, none of the Intellectual Property is being infringed by

third parties. Except as set forth on Schedule 3.1.19(e), there is no claim or demand of any Person pertaining to, or any proceeding which is pending or, to the knowledge of Seller, threatened, that challenges the rights of Seller in respect of any Intellectual Property, or claims that any default exists under any Intellectual Property License. None of the Owned Intellectual Property or, to the knowledge of Seller, the Intellectual Property Licenses is subject to any outstanding order, ruling, decree, judgment or stipulation by or with any court, tribunal arbitrator, or other Governmental Authority. The Owned Intellectual Property has been duly registered with, filed in or issued by, as the case may be, the United States Patent and Trademark Office and United States Copyright Office or other filing offices, domestic or foreign, to the extent commercially necessary or commercially desirable and the same remain in full force and effect.

3.1.20. Insurance. Schedule 3.1.20 contains a complete and correct list and summary description of all insurance policies maintained by Seller for the benefit of or in connection with the Assets. Seller has delivered to Buyer complete and correct copies of all such policies together with all riders and amendments thereto. Such policies are in full force and effect, and all premiums due thereon have been paid. Seller has complied in all material respects with the terms and provisions of such policies. The insurance coverage provided by such policies is adequate and customary. Schedule 3.1.20 sets out all claims made by Seller under any policy of insurance during the past two years and in the opinion of Seller reasonably formed and held, there is no basis on which a claim should or could be made under any such policy with respect to it.

3.1.21. Real Property.

- (a) Leases. Schedule 3.1.21(a) contains a complete and correct list of (i) all Leases setting forth the address, landlord and tenant for each Lease and (ii) all Other Leases, setting forth the address, landlord and tenant for each Other Lease. Seller has delivered to Buyer correct and complete copies of the Leases and the Other Leases. Each Lease and Other Lease is legal, valid, binding, enforceable, and in full force and effect, except as may be limited by bankruptcy, insolvency, reorganization and similar Applicable Laws affecting creditors generally and by the availability of equitable remedies. Neither Seller nor any other party is in default, violation or breach in any respect under any Lease or Other Lease, and no event has occurred and is continuing that constitutes or, with notice or the passage of time or both, would constitute a default, violation or breach in any respect under any Lease or Other Lease. Each Lease grants the tenant under the Lease the exclusive right to use and occupy the demised premises thereunder. Seller has good and valid title to the leasehold estate under each Lease free and clear of all Liens other than Permitted Liens. Seller enjoys peaceful and undisturbed possession under its respective Leases for the Real Property.
- (b) Fee and Leasehold Interests, etc. Other than the Plum Facility and the Murrysville Facility, the Real Property constitutes all the fee and leasehold interests in real property held for use in connection with, necessary for the conduct of, or otherwise material to, Seller's business.
- (c) No Proceedings. There are no eminent domain or other similar proceedings pending or threatened affecting any portion of the Real Property. There is no writ, injunction, decree, order or judgment outstanding, nor any action, claim, suit or proceeding, pending or threatened, relating to the ownership, lease, use, occupancy or operation by any Person of any Real Property.
- (d) Current Use. The use and operation of the Real Property by Seller does not violate in any material respect any instrument of record or agreement affecting the Real Property. To the knowledge of Seller, there is no violation of any covenant, condition, restriction, easement or order of any Governmental Authority having jurisdiction over such property or of any other Person entitled to enforce the same affecting the Real Property or the use or occupancy thereof. No damage or destruction has occurred with respect to any of the Real Property since December 31, 2003 that would, individually or in the aggregate, have a Material Adverse Effect.
- (e) Compliance with Real Property Laws. The Real Property is in full compliance with all applicable building, zoning, subdivision and other land use and similar Applicable Laws affecting the Real Property (collectively, the "Real Property Laws"), and Seller has received no notice of violation or claimed violation of any Real Property Law. There is no pending or to the knowledge of Seller, anticipated change in any Real Property Law that will have or result in a material adverse effect upon the ownership, alteration,

use, occupancy or operation of the Real Property or any portion thereof. No current use by Seller of the Real Property is dependent on a nonconforming use or other Governmental Approval the absence of which would materially limit the use of such properties or assets held for use in connection with, necessary for the conduct of, or otherwise material to, Seller's business. 3.1.22. Environmental Matters.

- (a) Permits. All Environmental Permits are identified in Schedule 3.1.22(a), and Seller currently holds, and at all times has held, all such Environmental Permits necessary to conduct its business, and all such Environmental Permits shall be validly transferred to Buyer on the Closing Date. Seller has not been notified by any relevant Governmental Authority that any Environmental Permit will be modified, suspended, cancelled or revoked, or cannot be renewed in the ordinary course of business without any material change in any of the terms or conditions.
- (b) No Violations. Seller has complied and is in compliance in all respects with all Environmental Permits and all applicable Environmental Laws. No Person has alleged any violation by Seller of any Environmental Permits or any applicable Environmental Law relating to the conduct of its business or the use, ownership or transferability of the Real Property. (c) No Actions. Except as set forth in Schedule 3.1.22(c), Seller has not caused or taken any action that has resulted or may result in, or has
- has not caused or taken any action that has resulted or may result in, or has been or is subject to, any liability or obligation relating to (i) the environmental conditions on, at, under, from or about any Real Property, the Assets or other properties or assets owned, leased or used by Seller or (ii) the past or present use, management, handling, transport, treatment, generation, storage or Release of any Hazardous Substances, except for any such liabilities and obligations that, individually and in the aggregate, are not material to Seller and have not had or resulted in, and will not have or result in, a Material Adverse Effect.
- (d) Other. Except as set forth in Schedule 3.1.22(d):
- (i) None of the current or past operations, or any by-product thereof, and none of the currently or formerly owned property or assets of Seller, including without limitation the Assets and the Real Property, is related to or subject to any investigation or evaluation by any Governmental Authority, as to whether any Remedial Action is needed to respond to a Release or threatened Release of any Hazardous Substances.
- (ii) Seller is not subject to any outstanding order, judgment, injunction, decree or writ from, or contractual or other obligation to or with, any Governmental Authority or other Person in respect of which Buyer may be required to incur any Environmental Liabilities and Costs arising from the Release or threatened Release of a Hazardous Substance.
- (iii) None of the Real Property is, and Seller has not transported or arranged for transportation (directly or indirectly) of any Hazardous Substances relating to the Assets or the Real Property to any location that is, listed or proposed for listing under CERCLA, or on any similar state list, or the subject of federal, state or local enforcement actions or investigations or Remedial Action.
- (iv) No work, repair, construction or capital expenditure is required or planned in respect of the Assets pursuant to or to comply with any Environmental Law, nor has Seller received any notice of any such requirement, except for such work, repair, construction or capital expenditure as is not material to the ordinary course of business.
 (v) The Seller has not received written notice of any past, present or future events, conditions, circumstances, activities, practices, incidents, actions or plans which may interfere with or prevent continued compliance by the Seller with applicable Environmental Laws, or which may
- compliance by the Seller with applicable Environmental Laws, or which may give rise to any common law or legal liability, or otherwise form the basis of any claim, action, suit, proceeding, hearing or investigation under applicable Environmental Laws, the subject of which is unresolved and which would have a Material Adverse Effect.
- (vi) There are no Hazardous Substances or underground storage tanks in, on, or under any real property owned or leased by the Seller as of the date hereof, except those that are both (i) in compliance with all applicable Environmental Laws and Environmental Permits and (ii) fully disclosed to Buyer in writing.
- (vii) There are no past, present or threatened Releases of Hazardous Substances in, on, under or from any real property owned or leased by the Seller.
- (viii) To the best knowledge of the Seller, there is no Release or threat of any Release of Hazardous Substances migrating to any real property owned or leased by the Seller.
- (ix) To the best of the knowledge of the Seller, there has been

no Release of Hazardous Substances for which Seller could have any liability under any applicable Environmental Law at any real property formerly owned, operated or leased by the Seller.

- (x) There is no statute or regulation which requires any disclosure of environmental conditions to any Governmental Authority or to Buyer, or any deed notice or other filing as a result of the contemplated transaction.
- (e) Full Disclosure. Seller has disclosed and made available to Buyer all information, including without limitation all studies, analyses and test results, in the possession, custody or control of Seller relating to (i) the environmental conditions on, under or about the Real Property, and (ii) Hazardous Substances used, managed, handled, transported, treated, generated, stored or Released by Seller or any other Person at any time on any Real Property, or otherwise in connection with the use or operation of the properties or assets of Seller.
- 3.1.23. Employees, Labor Matters, etc. Except as set forth in Schedule 3.1.23, Seller is not a party to or bound by any collective bargaining agreement and there are no labor unions or other organizations representing. purporting to represent or attempting to represent any employees employed by Seller. Since December 31, 2003 there has not occurred or, to the best knowledge of Seller after due inquiry, been threatened any material strike, slowdown, picketing, work stoppage, concerted refusal to work overtime or other similar labor activity with respect to any employees employed by Seller. There are no labor disputes currently subject to any grievance procedure, arbitration or litigation and there is no representation petition pending or, to the best knowledge of Seller after due inquiry, threatened with respect to any employee employed by Seller. Seller has complied with all provisions of Applicable Law pertaining to the employment of employees, including, without limitation, all such Laws relating to labor relations, equal employment, fair employment practices, entitlements, prohibited discrimination or other similar employment practices or acts, except for any failure so to comply that, individually or together with all such other failures, has not and will not result in a material liability or obligation on the part of Buyer, and has not had or resulted in, and will not have or result in, a Material Adverse Effect.
- 3.1.24. Employee Benefit Plans and Related Matters.
- (a) Employee Benefit Plans. Schedule 3.1.24(a) sets forth a true and complete list of each "employee benefit plan" as such term is defined in section 3(3) of the ERISA, whether or not subject to ERISA, and each bonus, incentive or deferred compensation, severance, termination, retention, change of control, stock option, stock appreciation, stock purchase, phantom stock or other equity-based, performance or other employee or retiree benefit or compensation plan, program, arrangement, agreement, policy or understanding, whether written or unwritten, that provides or may provide benefits or compensation in respect of any employee or former employee employed or formerly employed by Seller or the beneficiaries or dependents of any such employee or former employee (such employees, former employees, beneficiaries and dependents collectively, the ("Employees") or under which any Employee is or may become eligible to participate or derive a benefit and that is or has been maintained or established by Seller or any other trade or business, whether or not incorporated, which, together with Seller is or would have been at any date of determination occurring within the preceding six years treated as a single employer under Section 414 of the Code (such other trades and business collectively, the "Related Persons"), or to which Seller or any Related Person contributes or is or has been obligated or required to contribute or with respect to which Seller may have any liability or obligation (collectively, the "Plans"). With respect to each such Plan, Seller has provided Buyer complete and correct copies of: all written Plans; descriptions of all unwritten Plans; all trust agreements, insurance contracts or other funding arrangements; the two most recent actuarial and plan asset valuation reports; the most recent Forms 5500 and all schedules thereto; the most recent IRS determination letter; current summary plan descriptions; all material communications received from or sent to the IRS, the Pension Benefit Guaranty Corporation or the Department of Labor (including a written description of any oral communication); an actuarial study of any post-employment life or medical benefits provided under any such Plan, if any; statements or other communications regarding withdrawal or other multiemployer plan liabilities, if any; and all amendments and modifications to any such document. Seller has not communicated to any Employee any intention or commitment to modify any Plan or to establish or implement any other employee or retiree benefit or compensation arrangement.

401(a) of the Code, and the trust (if any) forming a part thereof, has received a favorable determination letter from the IRS as to its qualification under the Code and to the effect that each such trust is exempt from taxation under section 501(a) of the Code, and nothing has occurred since the date of such determination letter that could adversely affect such qualification or tax-exempt status.

- (c) Compliance; Liability.
- (i) No Plan, other than a Seller Pension Plan, is subject to section 412 of the Code or section 302 or Title IV of ERISA. As of the last day of the most recently ended fiscal year of the Seller Pension Plan, the "accumulated benefit obligations" (within the meaning of the Financial Accounting Standards Board Statement No. 87) under such Plan did not exceed the fair market value of the assets of such Plan allocable to such "accumulated benefit obligations," determined on the basis of the actuarial assumptions specified for financial statement purposes in the actuarial report prepared for such fiscal year of such Plan, each of which assumptions is reasonable.
- (ii) No liability has been or is expected to be incurred by Seller or any Related Person (either directly or indirectly, including as a result of an indemnification obligation) under or pursuant to Title I or IV of ERISA or the penalty, excise tax or joint and several liability provisions of the Code relating to employee benefit plans that could, following the Closing, become a liability of Buyer or of any employee benefit plan established or contributed to by Buyer and, to the best knowledge of Seller after due inquiry, no event, transaction or condition has occurred or exists that could result in any such liability to Seller or, following the Closing, Buyer.
- (iii) Each of the Plans has been operated and administered in all respects in compliance with all Applicable Laws, except for any failure so to comply that, individually or together with all other such failures, has not and will not result in a material liability or obligation on the part of Seller, or, following the Closing, Buyer, and has not had or resulted in, and will not have or result in, a Material Adverse Effect. There are no material pending or, to the best knowledge of Seller after due inquiry, threatened claims by or on behalf of any of the Plans, by any Employee or otherwise involving any such Plan or the assets of any Plan (other than routine claims for benefits).
- (iv) No Plan is a "multiemployer plan" within the meaning of Section 4001(a)(3) of ERISA or is a "multiple employer plan" within the meaning of section 4063 or 4064 of ERISA.
- (v) All contributions required to have been made by Seller and each Related Person to any Plan under the terms of any such Plan or pursuant to any applicable collective bargaining agreement or Applicable Law have been made within the earliest time prescribed by any such Plan, agreement or Applicable Law.
- (vi) No Transferred Employee is or may become entitled to postemployment benefits of any kind by reason of employment by Seller, including, without limitation, death or medical benefits (whether or not insured), other than (a) coverage provided pursuant to the terms of any Plan specifically identified as providing such coverage in Schedule 3.1.24(a) or mandated by section 4980B of the Code, (b) retirement benefits payable under any Plan qualified under section 401(a) of the Code or (c) deferred compensation accrued as a liability on the Balance Sheets or incurred after December 31, 2003 in the ordinary course of business consistent with the prior practice of Seller, pursuant to the terms of a Plan. The consummation of the transactions contemplated by this Agreement or the Collateral Agreements will not result in an increase in the amount of compensation or benefits or the acceleration of the vesting or timing of payment of any compensation or benefits payable to or in respect of any Transferred Employee. 3.1.25. Confidentiality. Except as set forth on Schedule 3.1.25, Seller has taken all steps necessary to preserve the confidential nature of all material confidential information (including, without limitation, any proprietary information) of Seller including but not limited to the manufacturing or marketing of any of Seller's products or services. 3.1.26. No Guarantees. Except as set forth on Schedule 3.1.26, none of the obligations or liabilities of Seller incurred in connection with the operation of its business is guaranteed by or subject to a similar contingent obligation of any other Person. Seller has not guaranteed or become subject to a similar contingent obligation in respect of the obligations or liabilities of any other Person. There are no outstanding letters of credit, surety bonds or similar instruments of Seller or any of its Affiliates in connection with the Assets.

3.1.27. Records. Except as set forth on Schedule 3.1.27, the minute books of Seller are substantially complete and correct in all material respects. The books of account of Seller are sufficient to prepare the Financial Statements in accordance with GAAP.

3.1.28. Brokers, Finders, etc. Except for the engagement of Parker/Hunter, Inc., the fees and expenses of which will be paid by Buyer, all negotiations relating to this Agreement, the Collateral Agreements, and the transactions contemplated thereby, have been carried on without the participation of any Person acting on behalf of Seller in such manner as to give rise to any valid claim against Buyer or any of its subsidiaries for any brokerage or finder's commission, fee or similar compensation, or for any bonus payable to any officer, director, employee, agent or sales representative of or consultant to Seller or their respective Affiliates upon consummation of the transactions contemplated hereby or thereby. 3.1.29. Disclosure. No representation or warranty by Seller contained in this Agreement nor any statement or certificate furnished or to be furnished by or on behalf of Seller to Buyer or its representatives in connection herewith or pursuant hereto contains or will contain any untrue statement of a material fact, or omits or will omit to state any material fact required to make the statements contained herein or therein not misleading. There is no fact (other-than matters of a general economic or political nature which do not affect Seller uniquely) known to Seller that has not been disclosed by Seller to Buyer that might reasonably be expected to have or result in a Material Adverse Effect.

3.1.30. Receivables. All of Seller's receivables (including accounts receivable, loans receivable and advances) which are reflected in the Financial Statements, and all such receivables which will have arisen since the Balance Sheet Dates, shall have arisen only from bona fide transactions in the ordinary course of business. Seller has no knowledge of any facts or circumstances generally (other than general economic conditions) which would result in any material increase in the uncollectability of such receivables as a class in excess of the reserves therefor set forth on the Financial Statements. Schedule 3.1.30 hereto accurately lists as of May 31, 2004 all receivables arising of Seller in excess of \$5,000, the amount owing and the aging of such receivable, the name and last known address of the party from whom such receivable is owing, and any security in favor of Seller for the repayment of such receivable which Seller purports to have. Seller has delivered to Buyer complete and correct copies of all instruments, documents and agreements evidencing such receivables and of all instruments, documents or agreements creating security therefor ("Security"). Seller has valid and perfected security interests in such Security (to the extent such priority may be obtained under applicable law by possession of such Security or the filing of financing statements or similar documents with respect thereto). 3.2. Representations and Warranties of Buyer and Matthews. Each of Buyer and Matthews represents and warrants to Seller as follows:

3.2.1. Corporate Status; Authorization, etc. Each of Buyer and Matthews is a corporation duly organized, validly existing and in good standing, under the laws of the Commonwealth of Pennsylvania with full corporate power and authority to execute and deliver this Agreement and the Collateral Agreements to which it is a party, to perform its obligations thereunder and to consummate the transactions contemplated hereby and thereby. The execution and delivery by such parties of this Agreement and the Collateral Agreements to which it is a party, and the consummation of the transactions contemplated hereby and thereby, have been authorized by all requisite corporate action of such parties. Such parties have duly executed and delivered this Agreement and the Collateral Agreements to which it is a party. This Agreement and each of the Collateral Agreements are the valid and legally binding obligations of such parties, enforceable against such parties in accordance with their respective terms.

3.2.2. No Conflicts, etc. The execution, delivery and performance by each of Buyer and Matthews of this Agreement and each of the Collateral Agreements to which it is a party, and the consummation of the transactions contemplated hereby and thereby, do not and will not conflict with or result in a violation of or under (with or without the giving of notice or the lapse of time, or both) (i) the articles of incorporation or by-laws or other organizational documents of such parties, (ii) any Applicable Law applicable to such parties or any of their properties or assets or (iii) any contract, agreement or other instrument applicable to such parties or any of their properties or assets, except, in the case of clause (iii), for violations and defaults that, individually and in the aggregate, have not and will not materially impair the ability of such parties to perform its obligations under this Agreement or under any of the Collateral Agreements to which it is a party. Except as specified in Schedule 3.2.2, no Governmental Approval or

other Consent is required to be obtained or made by such parties in connection with the execution and delivery of the this Agreement or the Collateral Agreements or the consummation of the transactions contemplated hereby and thereby.

- 3.2.3. Litigation. There is no action, claim, suit or proceeding pending, or to Buyer's or Matthews' knowledge threatened, by or against or affecting Buyer or Matthews in connection with or relating to the transactions contemplated by this Agreement or of any action taken or to be taken in connection herewith or the consummation of the transactions contemplated hereby.
- 3.2.4. Brokers, Finders, etc. Except for the engagement of Parker/Hunter, Inc., the fees and expenses of which will be paid by Buyer, all negotiations relating to this Agreement and the transactions contemplated hereby have been carried on without the participation of any Person acting on behalf of Buyer or Matthews in such manner as to give rise to any valid claim against Seller for any brokerage or finder's commission, fee or similar compensation.

ARTICLE IV COVENANTS

- 4.1. Covenants of Seller, iDL and BRR.
- 4.1.1. Further Assurances. Following the Closing, each of Seller, iDL and BRR Inc. shall, and shall cause each of its Affiliates to, from time to time, execute and deliver such additional instruments, documents, conveyances or assurances and take such other actions as shall be necessary, or otherwise reasonably requested by Buyer, to confirm and assure the rights and obligations provided for in this Agreement and in the Collateral Agreements and render effective the consummation of the transactions contemplated thereby.
- 4.1.2. Liability for Transfer Taxes. Seller shall be responsible for the timely payment of, and shall indemnify and hold harmless Buyer against, all sales (including, without limitation, bulk sales), use, value added, documentary, stamp, gross receipts, registration, transfer, conveyance, excise, recording, license and other similar Taxes and fees ("Transfer Taxes"), arising out of or in connection with or attributable to the transactions effected pursuant to this Agreement and the Collateral Agreements. Seller shall prepare and timely file all Tax Returns required to be filed in respect of Transfer Taxes (including, without limitation, all notices required to be given with respect to bulk sales taxes), provided that Buyer shall be permitted to prepare any such Tax Returns that are the primary responsibility of Buyer under applicable law. Buyer's preparation of any such Tax Returns shall be subject to Seller's approval, which approval shall not be withheld unreasonably.
- 4.1.3. Use of Business Name. After the Closing, each of Seller, iDL and BRR Inc. will not, directly or indirectly, do business, or allow any Affiliate to do business, or assist any third party in using or doing business, under the names and marks "Cloverleaf," "iDL" or "Big Red Rooster" (or any other name confusingly similar to such names and marks) except that Seller may continue to use the word "cloverleaf" (but not in connection with the word "group") as part of its legal name. Promptly after Closing iDL and BRR Inc. shall change their corporate names to not include "iDL" or "Big Red Rooster".
- 4.1.4. Access to Records. During the period in which Contingent Consideration is being calculated in accordance with Section 2.3, Buyer shall provide Seller and its representatives reasonable access during normal business hours to all of the books and records of Buyer to the extent that such access may reasonably be required by Seller in connection with matters relating to or affected by the operations of the business prior to the Closing. Seller shall be solely responsible for any costs or expenses incurred by it pursuant to this Section 4.1.4.
- 4.2. Covenants of Buyer.
- 4.2.1. Further Assurances. Following the Closing, Buyer shall, and shall cause its Affiliates to, from time to time, execute and deliver such additional instruments, documents, conveyances or assurances and take such other actions as shall be necessary, or otherwise reasonably requested by Seller, to confirm and assure the rights and obligations provided for in this Agreement and in the Collateral Agreements and render effective the consummation of the transactions contemplated thereby.
- 4.3. Operation of Business. The following covenants in Section 4.3 shall remain in effect during the period in which Contingent Consideration is being calculated in accordance with Section 2.3.
- 4.3.1. Operation of Business. After Closing, Buyer will consolidate the Assets and Assumed Liabilities of Seller under the name "Cloverleaf Group, Inc." Buyer shall continue to operate as an operating unit using the names

Cloverleaf, iDL and Big Red Rooster during the term of the Contingent Payment Period. Buyer shall be required to adhere to the internal control standards and codes of conduct as defined by the Sarbanes-Oxley Act of 2002, established regulatory standards and published Matthews' policies. All financial and administrative functions of Buyer shall be required to report directly to Matthews' corporate administration.

- 4.3.2. Board of Directors. After Closing, the Board of Directors of Buyer shall consist of such persons as are designated by Matthews and shall initially consist of David Kelly as sole director.
- 4.3.3. Board of Advisors. After Closing, the operational management of Buyer will be directed by a "Buyer Board of Advisors," subject to the authority of the Board of Directors of Buyer. The Buyer Board of Advisors shall initially consist of Ralph Murray, the Chairman of Buyer; Martin Beck, the President and Chief Executive Officer of Buyer; and David M. Kelly, the Chief Executive Officer of Matthews. The Buyer Board of Advisors shall be responsible for: (a) reviewing and assisting in establishing the strategies and goals for Buyer and assuring their achievement; (b) evaluating and recommending capital requests, including acquisitions; (c) reviewing and assisting in the formulation of Buyer's operating forecasts and budgets; and (d) identifying areas where Matthews' existing administrative, manufacturing, operational and distribution resources may be leveraged to result in synergistic savings for Buyer.
- 4.3.4. Newco Management Team. After Closing, the daily operational management of Buyer shall reside with the "Buyer Management Team." The President and CEO of Buyer shall be appointed by the Board of Directors. The initial President and CEO of Buyer shall be Martin Beck. The Buyer Management Team shall be chosen by the President of Buyer with review and approval thereof by the Chief Executive Officer of Matthews. The Buyer Management Team shall be responsible for: (a) developing the strategic focus and organizational culture of Buyer; (b) evaluating internal and acquisition growth strategies for Buyer; (c) developing operating budgets and forecasts for Buyer to be presented and reviewed at annual strategic management meetings; (d) formulating, evaluating and conveying capital requests to Matthews; (e) reporting financial and operating results on a monthly basis to Matthews; (f) evaluating and reviewing key personnel decisions; and (g) meeting regularly with Matthews' management to discuss progress. 4.3.5. Approval Process. In accordance with the requirements set forth herein, Buyer shall be required to submit to Matthews for review and
- (a) all individual capital expenditures of Buyer in excess of \$25,000, and annual capital expenditures in excess of \$100,000 in the aggregate;
- (b) all acquisitions to be made by Buyer;
- (c) all operating and capital budgets and forecasts of Buyer;
- (d) all changes to key management positions of Buyer, and the compensation and benefits of such key management positions;
- (e) operating leases with a net present value in excess of \$25,000;
- (f) disposition of any assets with a remaining net book value of more than \$25,000; and
- (g) the making of any charitable contributions.

Such approval process will be consistent with the approval process applied to Matthews' existing business units, as applicable. Notwithstanding the foregoing, (i) Matthews hereby agrees that approval of the items above shall not be withheld for reasons unrelated to the business of Buyer and (ii) Matthews management shall recommend to the Board of Directors of Matthews for approval such items so long as (x) with respect to items (a) and (e) above, the proposals meet the capital project and appropriation request criteria set forth on Schedule 4.3.5A hereto as determined by Matthews and (y) Buyer is achieving a 10% cumulative annual growth rate in Operating Profit from the consolidated operating profit of the Seller for the year end December 31, 2003, which the parties agree is \$4,208,000.

Matthews shall evaluate acquisitions at a cost of capital of 15%. The Buyer Management Team, in consultation with the Buyer Board of Advisors and Matthews, shall determine the financial characteristics and valuation matters of potential acquisition targets, in accordance with policies in existence from time to time of Matthews, which current policies are attached hereto as Schedule 4.3.5B.

4.3.6. Financing and Investment Arrangements. Buyer shall not be permitted to enter into any debt, capital lease or other financing arrangement with any entity other than Matthews. Buyer shall not be permitted to invest in any marketable securities.

4.3.7 BP Inventory. Buyer will warehouse and insure the BP Inventory at no

costs to iDL. Buyer will receive and fulfill usage orders from BP related to the BP Inventory at a cost to iDL equal to the amount currently being paid by BP for this service. Buyer will coordinate the billing for the BP Inventory, along with the related collection of such billings. Buyer will remit amounts collected with respect to the billings for the BP Inventory to iDL at no cost to iDL.

- 4.3.8. Acquisitions. During the Contingent Payment period, if Matthews intends to expand internally or make acquisitions in the line of business in which Buyer operates, it shall do so only through Buyer.
- 4.4. Shareholder Net Worth. None of the iDL Shareholders or the BRR Shareholders shall gift or transfer assets such that their personal net worth becomes less than the proceeds received by such shareholder in connection with the transactions contemplated by this Agreement, or, if greater, the personal net worth of such shareholder as of the Closing Date (including the proceeds of the transactions contemplated by this Agreement), for a period of eighteen (18) months from the Closing Date; provided, however, that the maximum net worth required to be maintained by any such shareholder shall be the maximum liability of such shareholder considering Section 7.2(g) hereof. ARTICLE V

CLOSING DELIVERIES

- 5.1. Deliveries of Seller. On the Closing Date in addition to the Assets, Seller shall deliver to Buyer the following:
- 5.1.1. Collateral Agreements. The following agreements (the "Collateral Agreements"):
- (a) the Escrow Agreement;
- (b) transitional lease agreements, pursuant to which iDL Inc. will provide to Buyer a lease of the Plum Facility and the Murrysville Facility.
- (c) An employment agreement with each of Ralph Murray and Martin Beck;
- (d) An employment agreement with each of the Key Employees.
- 5.1.2. Opinions of Counsel. An opinion, addressed to Buyer and dated the Closing Date, of Buchanan Ingersoll PC, special counsel to Seller and
- (ii) Williams Coulson Johnson Lloyd Parker & Tedesco LLC, each in substance and form reasonably satisfactory to Buyer.
- 5.1.3. Authorization. Certified resolutions of Seller, iDL Inc. and BRR Inc. approving this Agreement and the Collateral Agreements and all documents and instruments incident thereto.
- 5.1.4. Transfer Documents. The following transfer documents of Seller:
- (a) a bill of sale, assignment and general conveyance, in form and substance reasonably satisfactory to Buyer, dated the Closing Date, with respect to the Assets (other than any Asset to be transferred pursuant to any of the instruments referred to in any other clause of this Section 5.1.4);
- (b) an assignment of lease, dated as of the Closing Date, with respect to each Lease, and each Other Lease in form and substance reasonably satisfactory to Buyer, dated the Closing Date; and
- (c) certificates of title to all motor vehicles included in the Assets to be transferred to Buyer hereunder, duly endorsed for transfer to Buyer as of the Closing Date.
- 5.1.5. Consents and Estoppels. Buyer shall have received consents from the lessor of each Lease listed on Schedule 3.1.21(a) to the assignment of such Lease to Buyer. Buyer shall also have received estoppel certificates addressed to Buyer from the lessor of each Lease, identifying the Lease documents and any amendments thereto, stating that the Lease is in full force and effect and, to the best knowledge of the lessor, that the tenant is not in default under the Lease and no event has occurred that, with notice or lapse of time or both, would constitute a default by the tenant under the Lease and containing any other information reasonably requested by Buyer.
- 5.2. Deliveries of Buyer. On the Closing Date, Buyer shall deliver to Seller the following:
- 5.2.1. Opinion of Counsel. An opinion, addressed to Seller and dated the Closing Date, of Reed Smith LLP, counsel for Buyer and Matthews, in form and substance reasonably satisfactory to Seller.
- 5.2.2. Certified Resolutions. Certified resolutions of Buyer and Matthews approving this Agreement and the Collateral Agreements and the transactions contemplated thereby, and all documents and instruments incident thereto.
- 5.2.3. Collateral Agreements. Buyer shall have entered into each of the Collateral Agreements to which it is a party.

6.1. Employment of Seller's Employees.

(a) Seller will, and will cause each of its Affiliates to, use all reasonable efforts to cause the employees employed by Seller to make available their employment services to Buyer. For a period of two years from the Closing Date, Seller, iDL Inc. and BRR Inc. will not, and will not permit any of its Affiliates to, solicit, offer to employ or retain the services of or otherwise interfere with the relationship of Buyer with any Person employed by or otherwise engaged to perform services for Buyer. (b) Effective as of the Closing Date, Buyer shall offer employment to those employees selected by Buyer who are employed by Seller at compensation and vested benefits that are similar to those of such employees on the date hereof. Buyer shall offer employment to Ralph Murray and Martin Beck on the terms of the Employment Agreement described in connection herewith and to the Key Employees on the terms of the Employment Agreement described in connection herewith. Those employees who accept such offers of employment effective as of the Closing Date shall be referred to herein as the "Transferred Employees". Effective as of the Closing Date, Buyer shall assume the liability of Seller in respect of the Transferred Employees for accrued but unpaid salaries, wages, vacation and sick pay and incentive compensation, but only to the extent such liability is reflected on the Balance Sheets or relates to services rendered and arose after the Balance Sheet Dates in the ordinary course of business, consistent with the prior practice of Seller. Seller shall remain responsible for payment of any and all severance, retention, change in control or other similar compensation or benefits which are or may become payable in connection with the consummation of the transactions contemplated by this Agreement or the Collateral Agreements.

(c) Neither Buyer nor any of its Affiliates shall have any Benefit Liability with respect to any Employee or Plan or any claim thereof or related thereto except to the extent expressly provided in this Article VI with respect to the Transferred Employees. From and after the Closing, Seller shall, remain solely responsible for any and all Benefit Liabilities in respect of the Employees, including the Transferred Employees and their beneficiaries and dependents, relating to or arising in connection with or as a result of (i) the employment or the actual or constructive termination of employment of any such Employee by Seller (including, without limitation, in connection with the consummation of the transactions contemplated by this Agreement or the Collateral Agreements), (ii) the participation in or accrual of benefits or compensation under, or the failure to participate in or to accrue compensation or benefits under, any Plan or other employee or retiree benefit or compensation plan, program, practice, policy, agreement or arrangement of Seller or (iii) accrued but unpaid salaries, wages, bonuses, incentive compensation, vacation or sick pay or other compensation or payroll items (including, without limitation, deferred compensation), except, in any such case, to the extent any such Benefit Liability is (x) specifically assumed by Buyer pursuant to this Article VI or reflected on the Balance Sheets or relates to services rendered and arose after the Balance Sheet Dates in the ordinary course of business, consistent with the prior practice of Seller and in accordance with this Agreement. 6.2. Employee Benefit Plans, Welfare and Fringe Benefit Plans. (a)

Effective as of the Closing, Buyers shall assume each "employee benefit plan" in which Transferred Employees and their dependents and beneficiaries participate, as listed on Schedule 6.2, including any welfare and fringe benefit plans, programs, policies or arrangements established by Buyer for such Persons, provided that, from and after the Closing Date, Seller shall remain solely responsible for any and all Benefit Liabilities to or in respect of the Transferred Employees or their beneficiaries or dependents relating to or arising in connection with any claims, whether such claims are asserted before, on or after the Closing Date, for life, disability, accidental death or dismemberment, supplemental unemployment compensation, medical, dental, hospitalization, other health or other welfare or fringe benefits or expense reimbursements which claims relate to or are based upon an occurrence on or before the Closing Date (including claims for continuing treatment in respect of any illness, accident, disability, condition or confinement which occurs or commences on or before the Closing Date). (b) From and after the Closing, Seller shall remain solely responsible for any and all Benefit Liabilities in respect of any Employee (i) under any Plan that is an "employee welfare benefit plan" (within the meaning of section 3(1) of ERISA) that provides post-employment benefits of

any kind (a "Seller Retiree Welfare Plan"), other than as relates to Ronald D. Homitz pursuant to the Letter Agreement dated July 12, 2003 listed on Schedules 3.1.24(a) and 6.2 or (ii) otherwise in connection with the provision of, or the failure to provide, post-employment welfare benefits or

coverage to or in respect of any such Employee.

(c) From and after the Closing Date, Seller shall remain solely responsible for any and all Benefit Liabilities relating to or arising in connection with the requirements of section 4980B of the Code to provide continuation of health care coverage under any Plan in respect of (A) Employees, other than the Transferred Employees and their covered dependents, and (B) to the extent related to a qualifying event occurring on or before the Closing Date, Transferred Employees and their covered dependents.

6.3. Bonus Arrangements. After the Closing Date, Buyer shall provide certain Transferred Employees eligibility for participation in a bonus plan. The determination of which employees receive any such bonuses shall be made by the Buyer Management Team, subject to the approval of the President of Matthews. Any bonuses paid by Buyer shall be deducted from the Operating Profit of Buyer for purposes of Section 2.3.

6.4. Workers Compensation. From and after the Closing Date, Seller shall remain solely responsible for any and all Benefit Liabilities to or in respect of any Employee relating to or arising in connection with any and all claims for workers' compensation benefits arising in connection with any occupational injury or disease occurring or existing on or prior to the Closing Date.

6.5. Employment Taxes.

(a) Seller and Buyer will (i) treat Buyer as a "successor employer" and Seller as a "predecessor," within the meaning of Sections 3121(a)(1) and 3306(b) (1) of the Code, with respect to Transferred Employees who are employed by Buyer for purposes of Taxes imposed under the United States Federal Unemployment Tax Act ("FICA") or the United States Federal Insurance Contributions Act ("FUTA") and (ii) cooperate with each other to avoid, to the extent possible, the filing of more than one IRS Form W-2 with respect to each such Transferred Employee for the calendar year within which the Closing Date occurs.

(b) At the request of Buyer with respect to any particular applicable Tax law relating to employment, unemployment insurance, social security, disability, workers' compensation, payroll, health care or other similar Tax other than Taxes imposed under FICA and FUTA, Seller and Buyer will (i) treat Buyer as a successor employer and Seller as a predecessor employer, within the meaning of the relevant provisions of such Tax law, with respect to Transferred Employees who are employed by Buyer and (ii) cooperate with each other to avoid, to the extent possible, the filing of more than one individual information reporting form pursuant to each such Tax law with respect to each such Transferred Employee for the calendar year within which the Closing Date occurs.

ARTICLE VII

DEFINITIONS, MISCELLANEOUS

7.1. Definition of Certain Terms. The terms defined in this Section 7.1, whenever used in this Agreement (including in the Schedules), shall have the respective meanings indicated below for all purposes of this Agreement. All references herein to a Section, Article or Schedule are to a Section, Article or Schedule of or to this Agreement, unless otherwise indicated.

Adjusted Payment: as defined in Section 2.2.

Affiliate: of a Person means a Person that directly or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, the first Person. "Control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management policies of a person, whether through the ownership of voting securities, by contract or credit arrangement, as trustee or executor, or otherwise.

Agreement: this Asset Purchase Agreement, including the Schedules hereto.

Applicable Law: all applicable provisions of all (i) constitutions, treaties, statutes, laws (including the common law), rules, regulations, ordinances, codes or orders of any Governmental Authority, (ii) Governmental Approvals and (iii) orders, decisions, injunctions, judgments, awards and decrees of or agreements with any Governmental Authority.

Assets as defined in Section 1.1.

Assumed Liabilities: as defined in Section 2.5.

Audited Balance Sheet: the balance sheet contained in the

Audited Financial Statements.

Audited Balance Sheet Date: as defined in Section 3.1.4. Audited Financial Statements: as defined in Section 3.1.4.

Balance Sheets: the Audited Balance Sheet and the Unaudited

Balance Sheet, respectively.

Balance Sheet Dates: the Audited Balance Sheet Date and the

Unaudited Balance Sheet Date, respectively.

Benefit Liabilities: liabilities, obligations, commitments, costs and expenses, including reasonable fees and disbursements of attorneys and other advisors, including any such expenses incurred in connection with the enforcement of any applicable provision of this Agreement.

BRR Group: BRR Inc. and the BRR Shareholders.

Business Day: shall mean a day other than a Saturday, Sunday or other day on which commercial banks in New York City are authorized or required to close.

Buyer: as defined in the first paragraph of this Agreement.

Buyer Indemnitees: as defined in Section 7.2(a). Buyer's Accountants: PriceWaterhouse Coopers LLP.

Buyer Group: Buyer and Matthews.

CERCLA: the Comprehensive Environmental Response, Compensation

and Liability Act, as amended, 42 U.S.C. 9601 et seq.

Closing: as defined in Section 2.1. Closing Date: as defined in Section 2.1.

Closing Net Assets Statement: as defined in Section 2.4. Code: the Internal Revenue Code of 1986, as amended.

Collateral Agreements: the agreements and other documents and

instruments described in Sections 5.1.1 and 5.1.4.

Consent: any consent, approval, authorization, waiver, permit, grant, franchise, concession, agreement, license, exemption or order of, registration, certificate, declaration or filing with, or report or

notice to, any Person, including but not limited to any Governmental Authority.

Contract: as defined in Section 3.1.12(a). \$ or dollars: lawful money of the United States. Employees: as defined in Section 3.1.24(a).

Environmental Laws: all Applicable Laws relating to the protection of the environment, to human health and safety, or to any emission, discharge, generation, processing, storage, holding, abatement, existence, Release, threatened Release or transportation of any Hazardous Substances, including, without limitation, (i) CERCLA, the Resource Conservation and Recovery Act, and the Occupational Safety and Health Act, (ii) all other requirements pertaining to reporting, licensing, permitting, investigation or remediation of emissions, discharges, releases or threatened releases of Hazardous Substances into the air, surface water, groundwater or land, or relating to the manufacture, processing, distribution, use, sale, treatment, receipt, storage, disposal, transport or handling of Hazardous Substances, and (iii) all other requirements pertaining to the protection of the health and safety of employees or the public.

Environmental Liabilities and Costs: all Losses, whether direct or indirect, known or unknown, current or potential, past, present or future, imposed by, under or pursuant to Environmental Laws, including, without limitation, all Losses related to Remedial Actions, and all fees, disbursements and expenses of counsel, experts, personnel and consultants based on, arising out of or otherwise in respect of: (i) the ownership or operation of its business, Real Property or Other Leases or any other real properties, assets, equipment or facilities, by Seller, or any of their predecessors or Affiliates; (ii) the environmental conditions existing on the Closing Date on, at, under, from, above, or about any Real Property or property subject to Other Leases or any other real properties, assets, equipment or facilities currently or previously owned, leased or operated by the Seller, or any of their predecessors or Affiliates; and (iii) the transportation, disposal or other disposition prior to the Closing Date of any Hazardous Substances, other wastes or recycled or reclaimed materials generated by Seller; and (iv) expenditures necessary to cause any Real Property or any aspect of Seller's Business to be in compliance with any and all requirements of Environmental Laws as of the Closing Date, including, without limitation, all Environmental Permits issued under or pursuant to such Environmental Laws, and reasonably necessary to make full economic use of any Real Property.

Environmental Permits: any federal, state and local permit, license, registration, consent, order, administrative consent order, certificate, approval or other authorization with respect to Seller necessary for the conduct of its business as currently conducted or previously conducted under any Environmental Law.

ERISA: the Employee Retirement Income Security Act of 1974, as amended.

Excluded Assets: as defined in Section 1.2. Excluded Liabilities: as defined in Section 2.6.

Financial Statements: each of the financial statements required

to be provided by Section 3.1.4.

GAAP: generally accepted accounting principles as in effect in the United States.

Governmental Approval: any Consent of, with or to any Governmental Authority.

Governmental Authority: any nation or government, any state or other political subdivision thereof, any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, including, without limitation, any government authority, agency, department, board, commission or instrumentality of the United States, any State of the United States or any political subdivision thereof, and any tribunal or arbitrator(s) of competent jurisdiction, and any self-regulatory organization.

Hazardous Substances: any substance that: (i) is or contains asbestos, urea formaldehyde foam insulation, polychlorinated biphenyls, petroleum or petroleum derived substances or wastes, radon gas or related materials, lead-based paint or microbial matter, (ii) requires investigation, removal or remediation under any Environmental Law, or is defined, listed or identified as a "hazardous waste," "hazardous substance," "solid waste," "pollutant" or "contaminant" thereunder, or (iii) is toxic, explosive, corrosive, flammable, infectious, radioactive, carcinogenic, mutagenic, or otherwise hazardous and is regulated by any Governmental Authority or Environmental Law. iDL Group: iDL, Inc., Hugh Andrew, L.P. and the iDL Shareholders.

Indemnified Party: as defined in Section 7.2(e). Indemnifying Party: as defined in Section 7.2(e).

Intellectual Property: any and all United States and foreign: (a) patents (including design patents, industrial designs and utility models) and patent applications (including docketed patent disclosures awaiting filing, reissues, divisions, continuations-in-part and extensions), patent disclosures awaiting filing determination, inventions and improvements thereto; (b) trademarks, service marks, trade names, trade dress, logos, business and product names, slogans, and registrations and applications for registration thereof; (c) copyrights (including software) and registrations thereof; (d) inventions, processes, designs, formulae, trade secrets, know-how, industrial models, confidential and technical information, manufacturing, engineering and technical drawings, product specifications and confidential business information; (e) intellectual property rights similar to any of the foregoing; (f) copies and tangible embodiments thereof (in whatever form or medium, including electronic media).

Intellectual Property Assets: as defined in Section 1.1(h).

Inventories: as defined in Section 1.1(b).

IRS: the Internal Revenue Service.

Key Employees: C. Michael Dempe, Jim Krentz, Ken Bohl, David Ball Paul Kimmel, Rick Sheffel, Ben Urso, Aaron Spiess, Mike Hayes, Russ Branaghan and Diane Rambo.

Leases: means the real property leases, subleases, licenses and occupancy agreements pursuant to which either Seller is the lessee, sublessee, licensee or occupant.

Lien: any mortgage, pledge, hypothecation, right of others, claim, security interest, encumbrance, lease, sublease, license, occupancy agreement, adverse claim or interest, easement, covenant, encroachment, burden, title defect, title retention agreement, voting trust agreement, interest, equity, option, lien, right of first refusal, charge or other restrictions or limitations of any nature whatsoever, including but not limited to such as may arise under any Contracts.

Losses: as defined in Section 7.2(a).

Material Adverse Effect: any event, occurrence, fact, condition, change or effect that is materially adverse to the business, operations, prospects, results of operations, prospects, condition (financial or otherwise), properties (including intangible properties), assets (including intangible assets) or liabilities of Seller.

Multiemployer Plan: as defined in Section 3.1.24(c)(iv). Other Leases: the leases, subleases, licenses and occupancy agreements pursuant to which either Seller is a lessor, sublessor or licensor of any part of the Real Property.

Owned Intellectual Property: as defined in Section 3.1.19(a). Pension Plan: a pension plan as defined in section 3(2) of ERISA which is not an individual account plan as defined in section 3(34) of

Permitted Liens: (i) Liens reserved against in the Balance Sheets, to the extent so reserved, (ii) Liens for Taxes not yet due and payable or which are being contested in good faith and by appropriate proceedings if adequate reserves with respect thereto are maintained on Seller's books in accordance with GAAP, or (iii) Liens that, individually and in the aggregate, do not and would not materially detract from the value of any of the property or assets of Seller or materially interfere with the use thereof as currently used or contemplated to be used or otherwise.

Person: any natural person, firm, partnership, association, corporation, company, trust, business trust, Governmental Authority or other entity.

Plan: as defined in Section 3.1.24(a).

Prime Rate: the "Prime Rate" as set forth in the Wall Street

Journal from time to time.

Purchase Price: as defined in Section 2.2.

Real Property: all interests leased pursuant to the Leases. Real Property Laws: as defined in Section 3.1.21(e). Related Persons: as defined in Section 3.1.24(a). Release: any releasing, disposing, discharging, injecting,

spilling, leaking, leaching, pumping, dumping, emitting, escaping, emptying, seeping, dispersal, migration, transporting, placing and the like, including without limitation, the moving of any materials through, into or upon, any land, soil, surface water, ground water or air, or otherwise entering into the environment.

Remedial Action: all actions required to (i) clean up, remove, treat or in any other way remediate any Hazardous Substances; (ii) prevent the release of Hazardous Substances so that they do not

migrate or endanger or threaten to endanger public health or welfare or the environment; or (iii) perform studies, investigations and care (including any financial responsibility requirement) related to any such Hazardous Substances.

Security: as defined in Section 3.1.30.

Seller: as defined in the first recital of this Agreement.

Seller Indemnitees: as defined in Section 7.2(c).

Seller Group: the Seller, iDL Inc., Hugh Andrew, L.P., BRR Inc.,

the iDL Shareholders and the BRR Shareholders.

Subsidiaries: each corporation or other Person in which a Person owns or controls, directly or indirectly, capital stock or other equity interests representing at least 50% of the outstanding voting stock or

other equity interests.

Tax: any federal, state, provincial, local, foreign or other income, alternative, minimum, accumulated earnings, personal holding company, franchise, capital stock, net worth, capital, profits, windfall profits, gross receipts, value added, sales, use, goods and services, excise, customs duties, transfer, conveyance, mortgage, registration, stamp, documentary, recording, premium, severance, environmental (including taxes under Section 59A of the Code), real property, personal property, ad valorem, intangibles, rent, occupancy, license, occupational, employment, unemployment insurance, social security, disability, workers' compensation, payroll, health care, withholding, estimated or other similar tax, duty or other governmental charge or assessment or deficiencies thereof (including all interest and penalties thereon and additions thereto whether disputed or not). Tax Return: any return, report, declaration, form, claim for refund or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

Transaction Expenses: as defined in Section 7.4. Transferred Employees: as defined in Section 6.1(b).

Transfer Taxes: as defined in Section 4.1.2.

Treasury Regulations: the regulations prescribed pursuant to the

Code.

Unaudited Annual Balance Sheet: the balance sheet contained in the Unaudited Annual Financial Statements.
Unaudited Balance Sheet Date: as defined in Section 3.1.4.
Unaudited Annual Financial Statements: as defined in Section 3.1.4.

Withholding Taxes: as defined in Section 3.1.6(a). Working Capital Liabilities: as defined in Section 2.5(a).

7.2. Indemnification. (a) By iDL Group. Each of Seller and the iDL Group jointly and severally covenant and agree to defend, indemnify and hold harmless Buyer, its officers, directors, employees, agents, advisers, representatives and Affiliates (collectively, "Buyer Indemnitees") from and against, and pay or reimburse Buyer Indemnitees for, any and all claims, liabilities, obligations, losses, fines, costs, royalties, proceedings, deficiencies or damages (whether absolute, accrued, conditional or otherwise and whether or not resulting from third party claims), including out-of-pocket expenses and reasonable attorneys' and accountants fees incurred in the investigation or defense of any of the same or in asserting any of their respective rights hereunder (collectively, "Losses"), with respect to the

business or Assets which were previously owned by iDL, resulting from or arising out of:

- (i) any inaccuracy of any representation or warranty made by a member of the iDL Group or Seller herein or in any Collateral Agreement or in connection herewith or therewith;
- (ii) any failure of a member of the iDL Group or Seller to perform any covenant or agreement hereunder or any Collateral Agreement or fulfill any other obligation in respect hereof or of any Collateral Agreement;
- (iii) any Excluded Liabilities or Excluded Assets;
- (iv) the operation of the business by a member of the iDL Group or Seller or their ownership of the Assets prior to the Closing Date;
- (v) any and all Benefit Liabilities in respect of Employees, except, with respect to Transferred Employees, to the extent assumed by Buyer pursuant to Article VI;
- (vi) all Environmental Liabilities and Costs in any way related to iDL Group's or Seller's operation of the business or any environmental conditions in, on, under, from or about the Real Property, the Assets or other properties or assets owned, lease or used by Seller which were created, existed or arose prior to the Closing Date or relating to the Excluded Assets;
- (vii) any product liability claim with respect to products manufactured or sold or events occurring prior to the Closing; (viii) any failure of Seller to comply with applicable bulk sales laws (in consideration of which indemnification obligation Buyer hereby waives compliance by Seller with any applicable bulk sales laws); and (ix) any claims by a landlord with respect to any Lease of the iDL Group related to the failure to obtain a required consent to the transactions contemplated hereby or in any Collateral Agreement.
- (b) By BRR Group. Each of Seller and the BRR Group jointly and severally covenant and agree to defend, indemnify and hold harmless the Buyer Indemnitees, from and against, and pay or reimburse Buyer Indemnitees for, any and all claims, liabilities, obligations, losses, fines, costs, royalties, proceedings, deficiencies or damages (whether absolute, accrued, conditional or otherwise and whether or not resulting from third party claims), including out-of-pocket expenses and reasonable attorneys' and accountants fees incurred in the investigation or defense of any of the same or in asserting any of their respective rights hereunder (collectively, "Losses"), with respect to the business or Assets which were previously owned by BRR Inc., resulting from or arising out of:
- (i) any inaccuracy of any representation or warranty made by a member of the BRR Group or Seller herein or in any Collateral Agreement or in connection herewith or therewith;
- (ii) any failure of a member of the BRR Group or Seller to perform any covenant or agreement hereunder or any Collateral Agreement or fulfill any other obligation in respect hereof or of any Collateral Agreement;
- (iii) any Excluded Liabilities or Excluded Assets;
- (iv) the operation of the business by a member of the BRR Group or Seller or their ownership of the Assets prior to the Closing Date;
- (v) any and all Benefit Liabilities in respect of Employees,
- except, with respect to Transferred Employees, to the extent assumed by Buyer pursuant to Article VI:
- (vi) all Environmental Liabilities and Costs in any way related to BRR Group's or Seller's operation of the business or any environmental

conditions in, on, under, from or about the Real Property, the Assets or other properties or assets owned, lease or used by Seller which were created, existed or arose prior to the Closing Date or relating to the Excluded Assets:

- (vii) any product liability claim with respect to products manufactured or sold or events occurring prior to the Closing; (viii) any failure of Seller to comply with applicable bulk sales laws (in consideration of which indemnification obligation Buyer hereby waives compliance by Seller with any applicable bulk sales laws); and (ix) any claims by a landlord with respect to any Lease of the BRR Group related to the failure to obtain a required consent to the transactions contemplated hereby or in any Collateral Agreement.
- (c) By Buyer. The members of the Buyer Group jointly and severally covenant and agree to defend, indemnify and hold harmless Seller and their officers, directors, employees, agents, advisors, representatives and Affiliates (collectively, the "Seller Indemnitees") from and against any and all Losses resulting from or arising out of:
- (i) any inaccuracy in any representation or warranty made by the members of the Buyer Group herein or in any Collateral Agreement or in connection herewith or therewith; or
- (ii) any failure of the members of the Buyer Group to perform any covenant or agreement hereunder or any Collateral Agreement or fulfill any other obligation in respect hereof or any Collateral Agreement;
- (iii) the Assumed Liabilities;
- (iv) the operation of the business by the members of the Buyer Group or Buyer's use of the Assets following the Closing Date, except, in the case of the clause (iv), to the extent such Losses result from or arise out

of the Excluded Liabilities or constitute Losses for which the Seller Group is required to indemnify Buyer Indemnitees under Section 7.2(a). (d) Adjustments to Indemnification Payments. Any payment made by the Seller Group to Buyer Indemnitees, on the one hand, or by the Buyer Group to Seller Indemnitees, on the other hand, pursuant to this Section 7.2 in respect of any claim (i) shall be net of any insurance proceeds realized by and paid to the Indemnified Party in respect of such claim and (ii) shall be (A) reduced by an amount equal to any Tax benefits attributable to such claim and (B) increased by an amount equal to any Taxes attributable to the receipt of such payment, but only to the extent that such Tax benefits are actually realized, or such Taxes are actually paid, as the case may be, by the respective Indemnified Party or by any consolidated, combined or unitary group of which it is a member. The Indemnified Party shall use its reasonable efforts to make insurance claims relating to any claim for which it is seeking indemnification pursuant to this Section 7.2; provided that the Indemnified Party shall not be obligated to make such an insurance claim if the Indemnified Party in its reasonable judgment believes that the cost of pursuing such an insurance claim together with any corresponding increase in insurance premiums or other chargebacks to the Indemnified Party, as the case may be, would exceed the value of the claim for which the Indemnified Party is seeking indemnification.

(e) Indemnification Procedures. In the case of any claim asserted by a third party against a party entitled to indemnification under this Agreement (the "Indemnified Party"), notice shall be given by the Indemnified Party to the party required to provide indemnification (the "Indemnifying Party") promptly after such Indemnified Party has actual knowledge of any claim as to which indemnity may be sought, and the Indemnified Party shall permit the Indemnifying Party (at the expense of such Indemnifying Party) to assume the defense of any claim or any litigation resulting therefrom, provided that (i) the counsel for the Indemnifying Party who shall conduct the defense of such claim or litigation shall be reasonably satisfactory to the Indemnified Party, (ii) the Indemnified Party may participate in such defense at such Indemnified Party's expense, and (iii) the omission by any Indemnified Party to give notice as provided herein shall not relieve the Indemnifying Party of its indemnification obligation under this Agreement except to the extent that such omission results in a failure of actual notice to the Indemnifying Party and such Indemnifying Party is materially damaged as a result of such failure to give notice. Except with the prior written consent of the Indemnified Party, no Indemnifying Party, in the defense of any such claim or litigation, shall consent to entry of any judgment or enter into any settlement that provides for injunctive or other nonmonetary relief affecting the Indemnified Party or that does not include as an unconditional term thereof the giving by each claimant or plaintiff to such Indemnified Party of a release from all liability with respect to such claim or litigation. In the event that the Indemnified Party shall in good

faith determine that the conduct of the defense of any claim subject to indemnification hereunder or any proposed settlement of any such claim by the Indemnifying Party might be expected to affect adversely the Indemnified Party's Tax liability or the ability of Buyer to conduct its business, or that the Indemnified Party may have available to it one or more defenses or

counterclaims that are inconsistent with one or more of those that may be available to the Indemnifying Party in respect of such claim or any litigation relating thereto, the Indemnified Party shall have the right at all times to take over and assume control over the defense, settlement, negotiations or litigation relating to any such claim at the sole cost of the Indemnifying Party, provided that if the Indemnified Party does so take over and assume control, the Indemnified Party shall not settle such claim or litigation without the written consent of the Indemnifying Party, such consent not to be unreasonably withheld. In the event that the Indemnifying Party does not accept the defense of any matter as above provided, the Indemnified Party shall have the full right to defend against any such claim or demand and shall be entitled to settle or agree to pay in full such claim or demand. In any event, the Indemnifying Party and the Indemnified Party shall cooperate in the defense of any claim or litigation subject to this Section 7.2 and the records of each shall be available to the other with respect to such defense.

- (f) Time Limitation. All claims for indemnification under clause (i) of the first sentence of Section 7.2(a), 7.2(b) or 7.2(c) must be asserted within 30 days of the termination of the respective survival periods set forth in Section 7.3.
- (g) Dollar Limitation. The Indemnifying Party's liability for all losses in respect of claims made under: (A) Section 7.2(a)(i) and 7.2(a)(ii) shall be limited to \$13,770,000, (B) Section 7.2(b)(i) and 7.2(b)(ii) shall be limited to \$3,230,000; and (C) Section 7.2(c)(i) and 7.2(c)(ii) (other than any claim with respect to the failure to perform obligations related to any Contingent Consideration payments pursuant to Section 2.3) shall be limited to \$17,000,000; provided, however, that the foregoing limitation shall not apply to claims made with respect to Section 3.1.22. (h) Set-off.
- (i) Buyer shall be entitled to set-off against the Contingent Consideration Payments due to Ralph Murray, the amount of any outstanding claims made pursuant to this Section 7.2 with respect to iDL and Seller to the extent such claim is related to the business or net assets transferred by iDL.
- (ii) Buyer shall be entitled to set-off against the Contingent Consideration Payments due to Martin Beck, the amount of any outstanding claims made pursuant to this Section 7.2 with respect to BRR Inc. and Seller to the extent such claim is related to the business or net assets transferred by BRR Inc.
- (iii) Buyer shall not set off against the Contingent Consideration Payments that will ultimately be received by any partner of Seller in accordance with Section 2.3(c), the amount of any outstanding claims made pursuant to this Section 7.2, other than as permitted by subsections (a) and (b) above; provided, however, that the foregoing shall not affect the fact that the BRR Shareholders and iDL Shareholders are making representations and warranties hereunder and may be liable therefor. (i) Accounts Receivable Indemnity. The Seller Group jointly and severally shall indemnify the Buyer Indemnitees for all accounts receivable that were outstanding and recorded on the books of the Seller as of the Closing Date and remain uncollected as of January 1, 2006. The amount of indemnification will be reduced by the allowance for doubtful accounts recorded on the books of the Seller as of the Closing Date, with such allowance to be determined in accordance with generally accepted accounting principles and consistent with Seller's past accounting practices. Any accounts receivable indemnified by the Seller Group (including any credit insurance claim related to such specific receivables) will become the property of the Seller on the date such indemnification is paid. (j) Inventory Indemnity. The Seller Group jointly and severally shall indemnify the Buyer Indemnitees for all inventory items (raw materials, supplies, work-in-process, finished goods, etc.) that were on-hand (including consigned inventory, if any) and recorded on the books of the Seller as of the Closing Date and still on-hand as of January 1, 2006. The amount of indemnification will be equal to the book value of such inventory items as of the Closing Date and will be reduced by the reserve (if any) for slowmoving/obsolete inventory recorded on the books of the Seller as of the Closing Date, with such reserve to be determined in accordance with generally

accepted accounting principles and consistent with Seller's past accounting

practices. Any inventory indemnified by the Seller Group will become the property of the Seller on the date such indemnification is paid.

- (k) Calculation. For purposes of calculating the indemnification in subsections (i) and (j) above, the total uncollected accounts receivable and total obsolete inventory will be measured against the aggregate of the allowance for doubtful accounts and the reserve for slow-moving/obsolete inventory.
- 7.3. Survival of Representations and Warranties, etc. The representations and warranties contained in this Agreement shall survive the execution and delivery of this Agreement, any examination by or on behalf of the parties hereto and the completion of the transactions contemplated herein, but only to the extent specified below:
- (a) except as set forth in clauses (b) and (c) below, the representations and warranties contained in Section 3.1 and Section 3.2 shall survive for a period of eighteen months following the Closing Date;
- (b) the representations and warranties contained in Sections 3.1.1,
- 3.1.2, the first sentence of Section 3.1.11, and Section 3.2.1 shall survive without limitation; and
- (c) the representations and warranties contained in
- (i) Section 3.1.22 shall survive for a period of ten years following the Closing Date and (ii) Section 3.1.6 shall survive as to any Tax covered by such representations and warranties for so long as any statute of limitations for such Tax remains open, in whole or in part, including without limitation by reason of waiver of such statute of limitations.
- 7.4. Expenses. Except as otherwise provided herein, Seller, on the one hand, and Matthews, on the other hand, shall bear their respective expenses,

costs and fees (including attorneys' and auditors') in connection with the transactions contemplated hereby, including the preparation, execution and delivery of this Agreement and compliance herewith (the "Transaction Expenses"), whether or not the transactions contemplated hereby shall be consummated.

- 7.5. Severability. If any provision of this Agreement, including any phrase, sentence, clause, Section or subsection is inoperative or unenforceable for any reason, such circumstances shall not have the effect of rendering the provision in question inoperative or unenforceable in any other case or circumstance, or of rendering any other provision or provisions herein contained invalid, inoperative, or unenforceable to any extent whatsoever.
- 7.6. Notices. All notices, requests, demands, waivers and other communications required or permitted to be given under this Agreement shall be in writing and shall be deemed to have been duly given if (a) delivered personally, (b) mailed by first-class, registered or certified mail, return receipt requested, postage prepaid, or (c) sent by next-day or overnight mail or delivery or (d) sent by telecopy or telegram.

if to Buyer to,

Matthews International Corp.

Two NorthShore Center Pittsburgh, PA 15212 Attention: President Phone: (412) 442-8200 Facsimile: (412) 442-8290

with a copy to: Reed Smith LLP 435 Sixth Avenue Pittsburgh, PA 15219

Attention: Pasquale D. Gentile, Esq.

Phone: (412) 288-4112 Facsimile: (412) 288-3063

if to Seller,

Cloverleaf Group, L.P.

4250 Old William Penn Highway

Pittsburgh, PA 15146

Attention: Ralph W. Murray and Martin J. Beck

Facsimile: 412-380-4370

with a copy to: Buchanan Ingersoll PC One Oxford Centre 301 Grant Street, 20th Floor Attention: Carl A. Cohen, Esq.

Phone: (412) 562-8800 Facsimile: (412) 562-1041

or, in each case, at such other address as may be specified in writing to the other parties hereto.

All such notices, requests, demands, waivers and other communications shall be deemed to have been received (w) if by personal delivery on the day after such delivery, (x) if by certified or registered mail, on the seventh business day after the mailing thereof, (y) if by next-day or overnight mail or delivery, on the day delivered, (z) if by telecopy or telegram, on the next day following the day on which such telecopy or telegram was sent, provided that a copy is also sent by certified or registered mail.

7.7. Miscellaneous.

- 7.7.1. Headings. The headings contained in this Agreement are for purposes of convenience only and shall not affect the meaning or interpretation of this Agreement.
- 7.7.2. Entire Agreement. This Agreement (including the Schedules hereto) and the Collateral Agreements (when executed and delivered) constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, between the parties with respect to the subject matter hereof. Nothing contained in any Consent and Acknowledgement Agreement signed with any landlord of the Sellers shall be deemed to amend, waive, modify, or restrict any rights, obligations, representations or warranties of the parties set forth in this Agreement, and to the extent of any inconsistency between the provisions of such Consent and Acknowledgement and this Agreement, this Agreement shall control.
- 7.7.3. Counterparts. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall together constitute one and the same instrument.
- 7.7.4. Governing Law, etc. This Agreement shall be governed in all respects, including as to validity, interpretation and effect, by the internal laws of the Commonwealth of Pennsylvania, without giving effect to the conflict of laws rules thereof. The parties hereto hereby irrevocably submit to the jurisdiction of the courts of the Commonwealth of Pennsylvania and the Federal courts of the United States of America located in the Commonwealth and County of Allegheny solely in respect of the interpretation and enforcement of the provisions of this Agreement and of the documents referred to in this Agreement, and hereby waive, and agree not to assert, as a defense in any action, suit or proceeding for the interpretation or

enforcement hereof or of any such document, that it is not subject thereto or that such action, suit or proceeding may not be brought or is not maintainable in said courts or that the venue thereof may not be appropriate or that this Agreement or any of such document may not be enforced in or by said courts, and the parties hereto irrevocably agree that all claims with respect to such action or proceeding shall be heard and determined in such a Pennsylvania state or Federal court. The parties hereto hereby consent to and grant any such court jurisdiction over the person of such parties and over the subject matter of any such dispute and agree that mailing of process or other papers in connection with any such action or proceeding in the manner provided in Section 7.6, or in such other manner as may be permitted by law, shall be valid and sufficient service thereof.

- 7.7.5. Binding Effect. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, successors and permitted assigns.
- 7.7.6. Assignment. This Agreement shall not be assignable or otherwise transferable by any party hereto without the prior written consent of the other parties hereto, provided that Buyer may assign this Agreement to any subsidiary of Buyer.
- 7.7.7. No Third Party Beneficiaries. Except as provided in Section 7.2 with respect to indemnification of Indemnified Parties hereunder, nothing in this Agreement shall confer any rights upon any person or entity other than the parties hereto and their respective heirs, successors and permitted assigns.
- 7.7.8. Amendment; Waivers, etc. No amendment, modification or discharge of this Agreement, and no waiver hereunder, shall be valid or binding unless set forth in writing and duly executed by the party against whom enforcement of the amendment, modification, discharge or waiver is sought. Any such waiver shall constitute a waiver only with respect to the specific matter described in such writing and shall in no way impair the rights of the party granting such waiver in any other respect or at any other time. Neither the

waiver by any of the parties hereto of a breach of or a default under any of the provisions of this Agreement, nor the failure by any of the parties, on one or more occasions, to enforce any of the provisions of this Agreement or to exercise any right or privilege hereunder, shall be construed as a waiver of any other breach or default of a similar nature, or as a waiver of any of such provisions, rights or privileges hereunder. The rights and remedies herein provided are cumulative and are not exclusive of any rights or remedies that any party may otherwise have at law or in equity. The rights and remedies of any party based upon, arising out of or otherwise in respect of any inaccuracy or breach of any representation, warranty, covenant or agreement or failure to fulfill any condition shall in no way be limited by the fact that the act, omission, occurrence or other state of facts upon which any claim of any such inaccuracy or breach is based may also be the subject matter of any other representation, warranty, covenant or agreement as to which there is no inaccuracy or breach. The representations and warranties of the Seller Group shall not be affected or deemed waived by reason of any investigation made by or on behalf of Buyer (including but not

limited to by any of its advisors, consultants or representatives) or by reason of the fact that Buyer or any of such advisors, consultants or representatives knew or should have known that any such representation or warranty is or might be inaccurate. If any payment due hereunder shall be due on a day which is not a Business Day, such payment shall be made on the next succeeding Business Day.

7.7.9. Matters of Construction, Interpretation and the Like. The Buyer Group and the Seller Group have participated jointly in the negotiation and drafting of this Agreement. If an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Buyer Group and the Seller Group and no presumption or burden of proof shall arise favoring or disfavoring either the Buyer Group and the Seller Group because of the authorship of any of the provisions of this Agreement. Any reference to any United States Federal, state, local or foreign statute or law shall be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise. Unless the context of this Agreement otherwise requires, (a) words of any gender are deemed to include each other gender; (b) words using the singular or plural number also include the plural or singular number, respectively; (c) the terms "hereof," "herein," "hereby," "hereto," and derivative or similar words refer to this entire Agreement; (d) the terms "ARTICLE" or "Section" refer to the specified ARTICLE or Section of this Agreement; (e) the term "or" means "and/or"; (f) the term "party" means, on the one hand, the Buyer Group, on the other hand, the Seller Group, (g) the word "including" means "including without limitation"; and (h) all references to "dollars" or "\$" refer to currency of the United States of America. The exhibits and schedules specified in this Agreement are incorporated herein by reference and made a part hereof.

IN WITNESS WHEREOF, the parties have duly executed this Agreement as of the date first above written.

Cloverleaf Group, L.P.

By: Cloverleaf Group, LLC,
General Partner

By: Ralph W. Murray

Name: Ralph W. Murray

Title: Manager

By: Martin J. Beck

Name: Martin J. Beck

Title: Manager

	By: Ralph W. Murray
	Name: Ralph W. Murray
	Title: President
	Hugh Andrew, L.P.
	By: Helen Woodward, LLC, General Partner
	By: Ralph W. Murray
	Name: Ralph W. Murray
	Title: Manager
	Big Red Rooster Inc.
	By: Martin J. Beck
	Name: Martin J. Beck
	Title: President
	Cloverleaf Group, Inc. (Buyer)
	By: David M. Kelly
	Name: David M. Kelly
	Title: President
	Matthews International Corporation
	By: David M. Kelly
	Name: David M. Kelly
	Title: President
[additional signatures for	
[additional signatures it	5110W]
	IDL Shareholders
	Ralph W. Murray
	Ralph W. Murray
	Hugh A. Murray
	Hugh A. Murray
	Jeffrey H. Murray
	Jeffrey H. Murray
	Douglas G. Murray
	Douglas G. Murray

Kristy M. Holch	
Kristy M. Holch	
Cary M. Bohl	
Cary M. Bohl	
David G. Murray	
David G. Murray	
C. Michael Dempe	
C. Michael Dempe	
2000 Irrevocable Trust for Family of Ralph W. Murray	
By: Patricia P. Murray	
Patricia P. Murray, Trustee	
By: D. Grant Peacock D. Grant Peacock, Trustee	
2000 Irrevocable Trust for Family of Ralph W. Murray	
By: Elizabeth G. Murray	
Elizabeth G. Murray, Trustee	
BRR Shareholders	
Martin J. Beck	
Martin J. Beck	
Sweet Clover Group, Inc.	
By: Martin J. Beck	
Name: Martin J. Beck	
Title: President	

Continued from previous page Continued on following page

-2-

CERTIFICATION PRINCIPAL EXECUTIVE OFFICER

- I, David M. Kelly, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Matthews International Corporation;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and we have: a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
- b) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and c) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 10, 2004

David M. Kelly

_ _____

David M. Kelly Chairman of the Board, President and Chief Executive Officer

CERTIFICATION PRINCIPAL FINANCIAL OFFICER

- I, Steven F. Nicola, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Matthews International Corporation;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and we have: a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
- b) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- c) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 10, 2004

Steven F. Nicola

- -----

Steven F. Nicola Chief Financial Officer, Secretary and Treasurer

Exhibit 32.1

Certification Pursuant to 18 U.S.C. Section 1350,

As Adopted Pursuant to

Section 906 of The Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report of Matthews International Corporation (the "Company") on Form 10-Q for the period ending June 30, 2004 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, David M. Kelly, President and Chief Executive Officer, certify, to the best of my knowledge, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

David M. Kelly

- -----

David M. Kelly, President and Chief Executive Officer

August 10, 2004

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Matthews International Corporation and will be retained by Matthews International Corporation and furnished to the Securities and Exchange Commission or its staff upon request.

Exhibit 32.2

Certification Pursuant to 18 U.S.C. Section 1350,

As Adopted Pursuant to

Section 906 of The Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report of Matthews International Corporation (the "Company") on Form 10-Q for the period ending June 30, 2004 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Steven F. Nicola, Chief Financial Officer, certify, to the best of my knowledge, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Steven F. Nicola

- -----

Steven F. Nicola, Chief Financial Officer

August 10, 2004

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Matthews International Corporation and will be retained by Matthews International Corporation and furnished to the Securities and Exchange Commission or its staff upon request.